

Estudos e Documentos de Trabalho

Working Papers

16 | 2008

## IMPROVING COMPETITION IN THE NON-TRADABLE GOODS AND LABOUR MARKETS: THE PORTUGUESE CASE

Vanda Almeida Gabriela Castro Ricardo Mourinho Félix

October 2008

The analyses, opinions and findings of these papers represent the views of the authors, they are not necessarily those of the Banco de Portugal or the Eurosystem.

Please address correspondence to Ricardo Mourinho Félix Economics and Research Department Banco de Portugal, Av. Almirante Reis no. 71, 1150-012 Lisboa, Portugal; Tel.: 351 21 313 0321 , Email: rfelix@bportugal.pt

#### BANCO DE PORTUGAL

Economics and Research Department Av. Almirante Reis, 71-6th floor

1150-012 Lisboa www.bportugal.pt

#### Printed and distributed by

Administrative Services Department Av. Almirante Reis, 71-2nd floor 1150-012 Lisboa

## Number of copies printed 170 issues

Legal Deposit no. 3664/83 ISSN 0870-0117 ISBN 978-989-8061-52-2

## Improving competition in the non-tradable goods and labour markets: the Portuguese case<sup>\*</sup>

Vanda Almeida Gabriela Castro

Ricardo Mourinho Félix Economics and Research Department, Banco de Portugal<sup>†</sup>

October 21, 2008

#### Abstract

This study assesses the macroeconomic impacts of increasing competition in the nontradable goods and labour markets in Portugal. We lean on evidence that the maintenance of low competition in these markets may have contributed to the recent poor performance of the Portuguese economy. The analysis is performed using *PESSOA*, a dynamic general equilibrium model for a small-open economy integrated in a monetary union, featuring Blanchard-Yaari households, a multi-sectoral production structure and a number of nominal and real rigidities. We conclude that measures aimed at increasing competition in the Portuguese non-tradable goods and labour markets could induce important international competitiveness gains and be valuable instruments in promoting necessary adjustments within the monetary union framework. However, in the short run, real interest rates are likely to increase temporarily, driving consumption and output temporarily downwards.

Keywords: competition; competitiveness; DSGE; small-open economy; Portugal.

JEL classification numbers: E2, E60, F16, F41

<sup>&</sup>lt;sup>\*</sup>We would like to thank all the support provided by Douglas Laxton, Michael Kumhof and Antti Rippati. This paper also benefited from the comments and suggestions of António Antunes, João Amador, Isabel Horta Correia and participants at the 4th Dynare Conference and Banco de Portugal Research Department seminar. The views expressed in this paper are those of the authors and do not necessarily reflect the views of Banco de Portugal. Any errors and mistakes are ours.

<sup>&</sup>lt;sup>†</sup>Address for correspondence: Banco de Portugal, Departamento de Estudos Económicos, Rua Francisco Ribeiro, 2, 1150-165 Lisboa. E-mail addresses: vralmeida@bportugal.pt, ggcastro@bportugal.pt or rfelix@bportugal.pt

## **1** Introduction and motivation

In this study, we use  $PESSOA^1$ , a dynamic general equilibrium model for a small-open economy integrated in a monetary union, to assess the macroeconomic impact of reforms aimed at increasing competition in the Portuguese labour and non-tradable goods markets.

In the last 20 years, Portugal has been a case study for both good and bad reasons. During the 90s, it was frequently pointed out as a successful example of the European integration process. However, since the beginning of the current decade, the situation has changed significantly. In a context of a rapidly changing international environment, the Portuguese economy's structural fragilities have become evident. The country ceased to be an attractive destination for foreign direct investment, total factor productivity decelerated sharply and capital deepening went into a halt, leading to a slowdown of economic activity, to growth rates well below those of the euro area.

Two types of reasons are frequently put forward to explain the loss of competitiveness of the Portuguese economy. Firstly, the accession to the EU of Eastern Europe economies with lower wages, higher productivity levels, a more competitive tax system and sizeable EU transfers has diverted foreign direct investment away from Portugal. Secondly, the integration in international trade of emerging economies with very low unit labour costs and an export pattern similar to the Portuguese (mainly China), has led to significant market share losses, in particular in the textiles and clothing industry.<sup>2</sup> However, this reasoning can be disputed from at least two standpoints. Firstly, foreign direct investment, like every type of investment, flies to where it can yield expected returns higher than the cost of capital. One can thus hardly justify the decline of foreign direct investment inflows in Portugal using the fact that Eastern European countries benefited from it instead. More likely, the issue is that Portugal ceased to be a profitable destination for many industries in the current international context. Secondly, the EU as a whole largely benefited from the impact of opening its borders to exports from economies with low production costs, since this promoted a more efficient allocation of resources and expanded the consumption possibility frontier of the representative European household. The dual of this result is that inefficient Portuguese firms, previously sheltered from international competition, turned out to be unviable in the new international context. In our perspective, the fundamental issue is then: how to restore the competitiveness of Portuguese firms in the new international pattern of comparative advantages?

Blanchard and Giavazzi (2003) and Aghion, Askenazy, Bouelès, Cette and Dromel (2007) suggest that the level of product and labour market competition is an important factor behind the health of the business environment, which in turn determines to a large extent foreign direct investment inflows. Moreover, increasing competition is also a means to allow for better resource

<sup>&</sup>lt;sup>1</sup>*PESSOA* is the acronym for **P**ortuguese **E**conomy **S**tructural **S**mall **O**pen economy **A**nalytical model.

 $<sup>^{2}</sup>$ It is worth mentioning that the last stage of the multi-fiber agreement, which opened EU borders to textile imports from low labour cost economies, notably China, determined significant market share losses in European countries with an export pattern heavily dependent on these products, like Portugal and Italy.

allocation, promoting higher efficiency levels. In addition, as referred in Bayoumi, Laxton and Pesenti (2004) and in the European Commission (2008), higher competition promotes faster adjustment in the event of shocks, avoiding prolonged periods of anemic growth like the one the Portuguese economy has been experiencing since 2000. The available evidence (e.g. Conway and Nicoletti (2006), Høj, Jimenez, Maher, Nicoletti and Wise (2007), OECD (2006) and OECD (2008)) suggests that Portugal has a substantial margin to improve competition in its labour and non-tradable goods markets through the implementation of better regulation practices and a more active enforcement of the existing competition law. Leaning on these considerations and evidence, we suggest that increasing competition in the labour and non-tradable goods markets is a way to promote real exchange rate adjustment capable of addressing the current situation of Portugal within the EMU framework.

The use of a dynamic general equilibrium model to perform the analysis is crucial, since most of the impact of increasing competition in the non-tradable goods and labour markets is achieved through an increase in international competitiveness due to general equilibrium effects on tradable goods prices and on the real exchange rate. Moreover, dynamic general equilibrium models are currently the state of the art for simulation and policy analysis not only in academia, but also in policy-making institutions. Some interesting examples are: the use of the IMF's model, GIMF, presented in Kumhof and Laxton (2007b), for the analysis of the effects of fiscal policy in the US economy; the role played by the Sveriges Riksbank model, described in Adolfson, Laseén, Lindé and Villani (2005), in the context of policy analysis and forecasting; the extensive use of the AINO model, developed at the Bank of Finland, in the analysis of ageing and demographics in Kilponen, Kinnunen and Ripatti (2006a) and Kilponen, Kinnunen and Ripatti (2006b); and the New Area Model developed at the ECB and presented in Coenen, McAdam and Straub (2007).

The rest of the paper is structured as follows: section 2 surveys the empirical literature on product and labour market competition and positions Portugal in the context of the OECD economies; section 3 describes *PESSOA*; section 4 discusses its calibration; section 5 presents and discusses the simulation results; and finally, section 6 concludes and points some directions for further research.

# 2 The product and labour markets regulatory framework in Portugal

In the last years, several studies have addressed the macroeconomic impact of improving labour and product market competition. Bayoumi et al. (2004), using a general equilibrium model, found that significant impacts in output and employment may result from improving competition in the euro area, using a general equilibrium model. Kilponen and Ripatti (2006) found similar results for Finland, and Forni, Gerali and Pisano (2008) also provide the same assessment for Italy. European Commission (2005) assessed labour and product market regulatory framework in the EU and suggested that reforms aimed at increasing competition may substantially increase income and employment levels in the euro area.

According to product and labour market competition indicators available in OECD (2006) and OECD (2008) the Portuguese economy's relative position did not change much in the context of OECD countries, despite the important progress made between 1998 and 2003. These surveys suggest that reforms aimed at promoting greater competition in these markets would enhance the adjustment of the Portuguese economy to a new and more competitive international environment. On the product market front, it is suggested that the business sector environment can benefit a great deal from a reduction in the regulatory and administrative burden, which may lead to the entry of innovative firms, higher foreign direct investment inflows and more efficient resource allocation. Moreover, it is pointed out that the non-tradable goods sector, including network industries (e.g. electricity, gas and telecommunications) and retail distribution, requires procompetition regulation to increase efficiency and promote a decline of prices to levels closer to the marginal cost, since the market power of incumbents sharply limits competition.<sup>3</sup> Furthermore, high non-tradable goods prices have significant knock-on effects in tradable goods costs, hindering international competitiveness. On the labour market front, the evidence suggests a potentially dysfunctional labour market, exhibiting a steady increase in long-term unemployment and declining real wage flexibility. Labour demand is likely to be hindered by costs imposed by high employment protection legislation, while real wage adjustments turn out to be particularly difficult in a context of low inflation and low job creation.<sup>4</sup> Moreover, it is pointed out that unemployment benefits are extremely generous, implying higher reservation wages and contributing to longer unemployment spells. In this context, there is scope for measures aimed at promoting higher labour mobility and job creation and a tightening of unemployment benefits. These measures seem appropriate to improve competition in the labour market and introduce higher wage flexibility conditions.

The above-mentioned surveys are in line with Høj et al. (2007) stock-taking of product market regulation indicators for OECD countries.<sup>5</sup> As far as the Portuguese economy is concerned, this study refers to the fact that competition in services and, in particular, in network industries is hampered by state control on private business activities through special voting rights; in addition, the scope of administered prices in services is larger than in most OECD countries. This study

<sup>&</sup>lt;sup>3</sup>The information published in these surveys reveals that domestic and industrial electricity and gas pre-tax prices in Portugal are among the highest in OECD countries, despite some improvement in market liberalisation in the context of the Iberian Electricity Market. In the telecommunications sector, telephone charges declined towards OECD average in the mobile-phone sector, but fixed line charges for business remain high.

<sup>&</sup>lt;sup>4</sup>The available evidence suggests that the sensitivity of real wages to business cycle conditions is higher for newly hired workers than for workers that stay in the firm.

 $<sup>^{5}</sup>$ Conway, Janod and Nicoletti (2005) presents an updated set of product market regulation indicators for the whole economy and compares the 2003 situation of OECD countries with the 1998 situation.

also suggests that Portugal is likely to have significant price markups due to low competition in non-tradable goods as a result of weak antitrust indicators and a weak competition policy and law enforcement. Moreover, the tight planning regulations for large stores and restrictions in opening hours imply weak competition in the retail distribution sector.

Conway and Nicoletti (2006) measures cross-country differences in the regulatory framework of the non-tradable goods sector. The study stresses the importance of the non-tradable goods sector due to its weight in the economy and to the impact it has on consumers' welfare. Even more important is the fact that non-tradable goods prices exert a significant impact in tradable goods' marginal costs, not only because they are used as intermediate inputs, but also because of the impact they have in the price of investment goods and wages. The evidence found for Portugal indicates that the Portuguese regulatory framework in network industries (notably energy and telecommunications), retail distribution and professional services has room for improvement, since it is substantially restrictive in comparison with other OECD countries.

Høj (2007) takes stock of competition policy and law enforcement indicators and concludes that these vary substantially between countries. Portugal is pointed out as a case of weak competition policy and weak law enforcement, reflecting government special voting rights in private companies, restrictive entry regulations in network industries and low accountability of competition authorities.

Nicoletti and Scarpetta (2005) details evidence of the impact of product market reforms on employment and finds that product and labour market reforms may be complementary. Moreover, the authors find that restrictive product market regulations limit job creation, a point also noted in Blanchard and Giavazzi (2003), and that restrictive product regulations may interact with labour market settings that increase the bargaining power of the insiders, depressing employment further. The main policy implications are that large employment gains can be obtained by promoting higher product market competition in economies where this market is overly regulated and that the impacts on employment are likely to be more important in economies with more rigid labour markets.

Jean and Nicoletti (2002) uses cross-country data on industry-specific product market regulation and finds that it has a significant positive impact on wage premia due to the strategic interaction between management and workers in the sharing of monopolistic competition rents. Measures aimed at increasing competition in the product market therefore tend to reduce not only the goods price markup, but also the wage markup.

To sum up, the available evidence unambiguously suggests that Portugal can gather substantial benefits from implementing reforms aimed at improving labour and product market competition. These include enhanced flexibility, improved competition policies and better competition law enforcement. In the product market, this seems to be particularly true for the non-tradable goods sector. These reforms seem to be a powerful way to increase employment and income levels and promote a leap forward in the real convergence of the Portuguese economy within the euro area.

## 3 Introducing PESSOA

*PESSOA* is a model for a small open economy integrated in a fully-fledged monetary union. It features six types of economic agents: households, labour unions, manufacturers, distributors, the government and the rest of the world.

Households follow the Blanchard-Yaari overlapping generations model, which allows for non-Ricardian features of fiscal policy. <sup>6</sup> In addition, two types of households are considered, those which have perfect access to financial markets and those which are liquidity constrained. The first perform inter-temporal optimisation, are the sole owners of domestic firms and public debt stock (full home bias in asset detention is assumed) and are able to borrow from abroad, possibly at the cost of paying an exogenous risk premium on the monetary union interest rate.<sup>7</sup> The second can only perform intra-temporal optimisation, since they cannot save for the next period. This is a commonly adopted strategy to break the Ricardian equivalence in general equilibrium models and it is considered to be useful in obtaining realistic short-run responses to a fiscal policy rule adjustment. Since these households are not able to smooth consumption by adjusting portfolio, any shock affecting their budget constraint affects their consumption decisions immediately, ensuring that fiscal policy has immediate effects on output. Note that these households are not purely "ruleof-thumb" agents, since they have an optimisation plan, just like any other agent. All households consume and supply differentiated labour services.

Labour unions buy labour services from households and sell them to manufacturers at a higher price, exploiting the market power created by the fact that labour services are differentiated. The rents generated by this behaviour are then fully transferred to households, in the form of dividends. These are very particular dividends, since they simply correspond to the share of each household in the wedge generated by monopolistic competition in the labour market. This is a widely used modeling strategy to generate monopoly power and sticky wages in general equilibrium models, implying that households are rewarded in excess of their marginal disutility of labour.

Turning to the supply side, there are two types of firms: manufacturers, who produce two kinds of differentiated intermediate goods (tradables and non-tradables); and distributors, who produce four kinds of differentiated final goods (private consumption, government consumption, investment and export goods). All firms operate in monopolistic competition in their output markets and in perfect competition in their input markets. Manufacturers' technology features a deterministic

<sup>&</sup>lt;sup>6</sup>See original works from Blanchard (1985) and Yaari (1965). For a detailed discussion of this type of model see Blanchard and Fischer (1989) and Frenkel and Razin (1996). For applications to fiscal policy see Harrison, Nikolov, Quinn, Ramsay, Scott and Thomas (2005), Kilponen and Ripatti (2006) and Kumhof and Laxton (2007b)

<sup>&</sup>lt;sup>7</sup>In practice this allows the domestic interest rate to depart from the monetary union's interest rate.

labour-augmenting productivity trend, which is the only source of growth in the model.

The government has two main activities: consume and perform transfers across households through taxes and subsidies. To finance its activities, the government levies taxes on labour income, firm dividends and household consumption, and benefits from non-tax revenues stemming from EU transfers. Furthermore, it sells one-period government bonds to households, paying an interest rate on the stock of bonds held from one period to the next. To prevent an explosive debt path, a fiscal rule is imposed, according to which the labour income tax rate adjusts endogenously to ensure that the fiscal surplus-to-GDP ratio converges to a specified target value.

The rest of the world is assumed to correspond to the members of the monetary union (excluding the domestic economy), implying that the nominal effective exchange rate is irrevocably set to unity and that all flows are recorded in the same unit. Furthermore, since the domestic economy is small enough, domestic shocks are assumed to have no impact in the foreign economy. The two economies interact through both trade and financial flows. Where trade is concerned, this is done exclusively with distributors, to whom the rest of the world sells tradable intermediate goods and from whom it buys export goods. As for financial flows, since the home economy is small, changes in its net foreign position do not affect the foreign real interest rate.

The model includes wage and intermediate goods price markup shocks and a number of nominal and real frictions, such as price and wage rigidities, external habit formation, and investment and import content adjustment costs, which allow for smoother and more realistic short-term adjustment.

Some of conventions adopted in this study should be made clear from the outset. Firstly,  $p_t^X = \frac{P_t^X}{P_t}$  is the relative price of good X in terms of the numeraire price, which corresponds to the after-tax final consumption good price. Secondly, interest rates, inflation rates and risk premium are expressed as gross rates. Thirdly, the symbol \* indicates a foreign variable, that is, a euro area variable in the context of this model. Finally, we suppress the expectation operator for the sake of notation simplicity,  $x_{t+1} = E_t x_{t+1}$ ,  $\forall t$ .<sup>8</sup>

#### 3.1 Households

There are two types of overlapping generations optimising households (indexed by  $H \in \{OLG, LIQ\}$ ): OLG households, which have full access to financial markets; and LIQ households, which do not have access to financial markets, being thus forced to consume their after tax income in each period. There is a continuum of households (indexed by  $h \in (0,1)$ ), with each one supplying a different variety of labour.

<sup>&</sup>lt;sup>8</sup>The model exhibits a balanced growth path, therefore to render it stationary all model conditions must be divided by the determinist trend  $T_t$ . The model in the stationary form will be made available by the authors in a technical appendix.

In each period, a new generation of households is born. Each household faces an instant probability of death  $1-\theta$ , which is constant throughout life, irrespectively of age, and equal for all households. The probability of a household dying after t periods of life is then equal to  $(1-\theta)\theta^{t-1}$ , implying that its expected life horizon at any time is  $\frac{1}{1-\theta}$ .<sup>9</sup> Each generation is large enough so that  $1-\theta$  is also the decline rate of the generation's size over time. Thus, although each household is uncertain about the time of death, the size of a cohort declines deterministically through time, meaning that there is uncertainty for each individual household, but no aggregate uncertainty in this aspect. A convenient normalisation is that the size of a new generation of OLG and LIQ households is  $n(1-\psi)(1-\theta)$  and  $n\psi(1-\theta)$ , respectively, where  $\psi$  corresponds to the share of LIQ households in overall population. This implies that the number of households born in each period is  $n(1-\theta)$  and that the total number of households in each period is constant and equal to n.<sup>10</sup> Although this may seem a strong assumption, it is not completely at odds with the situation in many European countries, including Portugal.

A representative household h of type H with age a, derives utility from consumption,  $C_{a,t}^{H}(h)$ , relative to a consumption habit,  $Hab_{a,t}^{H}$ , and from leisure,  $1 - L_{a,t}^{H}(h)$ , assuming no bequest motive. The consumption habit is external, being a function of the lagged aggregate per capita consumption,  $C_{t-1}^H$ , which is unaffected by the household's h decisions, and is formulated as follows:

$$Hab_{a,t}^{OLG} = \left(\frac{C_{t-1}^{OLG}}{n(1-\psi)}\right)^v \quad \text{and} \quad Hab_{a,t}^{LIQ} = \left(\frac{C_{t-1}^{LIQ}}{n\psi}\right)^v \tag{1}$$

where  $0 \le v \le 1$  parameterises the degree of habit persistence.

The expected lifetime utility of each household at time t is given by:<sup>11</sup>

$$E_t \sum_{s=0}^{\infty} (\beta \theta)^s \frac{1}{1-\gamma} \left[ \left( \frac{C_{a+s,t+s}^H(h)}{Hab_{a+s,t+s}^H} \right)^{\eta^H} (1 - L_{a+s,t+s}^H(h))^{1-\eta^H} \right]^{1-\gamma}$$
(2)

where  $\gamma > 0$  is the coefficient of risk aversion  $(\frac{1}{\gamma})$  is the intertemporal elasticity of substitution),  $0 \le \eta^H \le 1$  is the consumption share parameter and  $0 \le \beta \le 1$  stands for the time discount factor. Note that each period's utility is discounted not only by the usual time discount factor,  $\beta$ , but also by  $\theta$ . This reflects the fact that households take into account that lifetime is a stochastic variable, since they know they will die, but do not know exactly when. Because of this, they attach an extra value to utility flows closer in time and therefore discount future utility more than implied by  $\beta$ .<sup>12</sup>

<sup>&</sup>lt;sup>9</sup>The expected life horizon can be obtained from the fact that  $\sum_{t=1}^{\infty} t(1-\theta)\theta^{t-1} = \frac{1}{(1-\theta)}$ . <sup>10</sup>Note that  $n(1-\theta) + n(1-\theta)\theta + n(1-\theta)\theta^2 + \dots = n(1-\theta)\sum_{s=1}^{\infty}\theta^{s-1} = n$ . <sup>11</sup>We consider the cashless limit economy advocated in Woodford (2003), and therefore, money demand is not included in the utility function.

 $<sup>^{12}</sup>$ This feature is known in the literature as over-discounting. For further details on this issue see Blanchard (1985), Frenkel and Razin (1996) and Harrison et al. (2005).

An important consequence of this is that it breaks the Ricardian equivalence, since households are not indifferent as to increases in government spending financed through current taxes or debt issuance (which implies future taxation). They strongly prefer debt, since they attach a positive probability to the event of not being alive by the time higher future tax payments are due.

The model features a life-cycle pattern following Blanchard (1985). Labour productivity of a household with age a,  $\Phi_{a,t}$ , is assumed to decline at a constant rate over lifetime, in order to generate a declining real wage pattern, in line with the formulation in Blanchard (1985):

$$\Phi_{a,t} = \Phi_a = k\chi^a, \quad \forall t \tag{3}$$

where  $0 \le \chi \le 1$  determines the labour productivity rate of decay. Note that this feature amplifies the non-Ricardian behaviour, since households prefer to be taxed tomorrow (when they are less productive and consequently their labour income is smaller) than today (when they are more productive, having a higher real wage). In addition, it should be mentioned that the formulation adopted is a very crude approximation to life-cycle behaviour, since in practice the labour income profile tends to be hump-shaped during working life and to collapse to zero at retirement age. However, in the context of the OLG Blanchard-Yaari model, analytical tractability concerning aggregation over the different generations is difficult to ensure with more sophisticated formulations. Furthermore, without loss of generality, the overall population's average productivity is normalised to unit, implying that the aggregate labour productivity of OLG and LIQ households is respectively  $n(1 - \psi)$  and  $n\psi$ .<sup>13</sup>

#### 3.1.1 OLG households

OLG households are distinguishable from LIQ households by the fact that they have access to financial markets and therefore can perform active consumption smoothing over time, minimising the impact of shocks to the economy on their consumption pattern.

An OLG household h with age a rents labour services to a union, receiving a wage for its services,  $W_t \Phi_a$  (reflecting its productivity level), and pays a labour income tax to the government at the tax rate  $\tau_{L,t}$ . Household h invests in financial assets, composed of domestic government bonds,  $B_{a,t}(h)$ , and foreign bonds,  $B_{a,t}^*(h)$ , which yield gross nominal interest rates  $i_t$  and  $i_t^*$ , respectively, on bonds held from period t to period t + 1. By convention, we assume that interest is paid at the

$$\frac{n(1-\psi)(1-\theta)\sum_{a=0}^{\infty}\theta^{a}\Phi_{a,t}}{n(1-\psi)} = 1 \Leftrightarrow n(1-\psi)(1-\theta)\sum_{a=0}^{\infty}\theta^{a}\Phi_{a,t} = n(1-\psi)$$
$$\frac{n\psi(1-\theta)\sum_{a=0}^{\infty}\theta^{a}\Phi_{a,t}}{n\psi} = 1 \Leftrightarrow n\psi(1-\theta)\sum_{a=0}^{\infty}\theta^{a}\Phi_{a,t} = n\psi$$

implying  $k = \frac{1-\theta\chi}{1-\theta}$ .

<sup>&</sup>lt;sup>13</sup>In order to ensure unit average productivity for OLG and LIQ households, we impose:

beginning of period t + 1. A risk premium wedge can be demanded for holding domestic assets,  $\Psi$ , reflecting the fact that domestic bonds may be riskier than foreign bonds. Furthermore, OLG households receive dividends from firms (indexed by  $D \in \{N, T, C, G, I, X\}$  i.e. non-tradable and tradable goods manufacturers and distributors of private consumption, government consumption, investment and export goods) and from unions. Since there is a continuum of firms (indexed by  $d \in (0, 1)$ ) and unions (indexed by  $h \in (0, 1)$ ) in the economy, we denote the after-tax nominal dividends received from firm d of type D by  $D_t^D(d, h)$  and the after-tax nominal dividends receives from union h by  $D_t^U(h)$ . In addition, the representative OLG household also receives lump-sum transfers from the government,  $TRG_t^{OLG}(h)$ , and from abroad,  $TRX_t^{OLG}(h)$ , and pays a lump-sum tax to the government,  $TRH_t(h)$ , which in turn redistributes it to LIQ households. This last type of transfer is a way of allowing LIQ households to benefit in some way from the existence of firm dividends without explicitly owning them, operating closely to an income insurance scheme.

Following the original work in Yaari (1965), the model features a life insurance company that operates under a zero-profit condition.<sup>14</sup> In a stochastic lifetime framework, in the absence of a life insurance company, households may die either as creditors, leaving unintended positive bequests, or as debtors, leaving unintended negative bequests. The insurance company solves the issue of unintended bequest by taking advantage of the fact that there is individual uncertainty but no aggregate uncertainty. All households sign a contract with the life insurance company, stating that they will receive a premium of  $\frac{1-\theta}{\theta}$  on each unit of their financial wealth for each period they survive, in exchange for their estates in the event of death.

The OLG household's budget constraint in nominal terms is then given by:

$$P_{t}C_{a,t}^{OLG}(h) + B_{a,t}(h) + B_{a,t}^{*}(h) = \frac{1}{\theta} \left[ i_{t-1}B_{a-1,t-1}(h) + i_{t-1}^{*}\Psi B_{a-1,t-1}^{*}(h) \right] + W_{t}\Phi_{a}L_{a,t}^{OLG}(h)(1-\tau_{L,t}) + \sum_{D=N,T,C,G,I,X} \int_{0}^{1} D_{a,t}^{D}(h,d)dd + D_{a,t}^{U}(h) - TRH_{a,t}(h) + TRG_{a,t}^{OLG}(h) + TRX_{a,t}^{OLG}(h)$$

$$(4)$$

The OLG household's optimisation problem is to choose consumption, labour supply, domestic bond holdings, and foreign bond holdings that maximise the present discounted value (PDV) of its future utility stream subject to the constraints imposed by (1), (3) and (4). The solution of this problem is a set of FOCs that express the optimal policy functions of the representative household of generation a. Since all OLG households within cohort a are identical, one can impose symmetric equilibrium, which implies that optimal conditions are identical for all households within cohort a, and therefore we can suppress from them the household's identifier h. From the combination of the

<sup>&</sup>lt;sup>14</sup>The zero-profit condition means that the insurer's receipts equal the payouts at each point in time, implying a large enough turnover of the population.

FOCs for consumption and labour supply, we obtain the consumption/leisure optimal allocation for households of generation a. This must be aggregated across the different age groups to obtain an aggregate consumption/leisure decision rule. For this, we need to aggregate consumption and labour, taking into account the size of each cohort at the time of birth,  $n(1 - \psi)(1 - \theta)$ , and the size of each of the remaining generations,  $n(1 - \psi)(1 - \theta)\theta^a$ :

$$C_t^{OLG} = n(1-\psi)(1-\theta) \sum_{a=0}^{\infty} \theta^a C_{a,t}^{OLG}$$
(5)

$$L_t^{OLG} = n(1-\psi)(1-\theta) \sum_{a=0}^{\infty} \theta^a (L_{a,t}^{OLG} \Phi_{a,t})$$
(6)

where  $C_t^{OLG}$  and  $L_t^{OLG}$  are respectively the aggregate level of consumption and labour supply of OLG households.

Applying this to the consumption/labour supply equation of cohort a and using the fact that aggregate productivity of OLG households is  $n(1 - \psi)$ , we get the following expression defining the aggregate consumption/labour supply decision rule of OLG households:

$$\frac{C_t^{OLG}}{n(1-\psi) - L_t^{OLG}} = \frac{\eta^{OLG}}{1 - \eta^{OLG}} w_t (1 - \tau_{L,t})$$
(7)

Combining the FOCs for domestic and foreign bonds, we obtain the optimal portfolio allocation, which yields the following interest rate parity condition:

$$i_t = i_t^* \Psi \tag{8}$$

which implies that the domestic interest rate is simply the risk-adjusted monetary union rate.

Finally, using the FOCs for consumption and domestic bonds, we obtain the following consumption Euler equation for a specific cohort of age a:

$$C_{a+1,t+1}^{OLG} = \left(\frac{C_t^{OLG}}{C_{t-1}^{OLG}}\right)^{v\eta^{OLG}(1-\frac{1}{\gamma})} \left(\frac{w_{t+1}\chi(1-\tau_{L,t+1})}{w_t(1-\tau_{L,t})}\right)^{(1-\eta^{OLG})(1-\frac{1}{\gamma})} (r_t\beta)^{\frac{1}{\gamma}} C_{a,t}^{OLG}$$
(9)

where  $r_t = \frac{i_t}{\pi_{t+1}}$  is the real interest rate and  $\pi_{t+1} = \frac{P_{t+1}}{P_t}$  stands for consumer price inflation rate.

Another fundamental equation is the consumption function, which relates the household's consumption to its wealth, through the marginal propensity to consume out of wealth  $(mpc_t)$ . To derive this function, we must impose a transversality condition on the household's optimisation problem, to ensure that expected asset holding at the time of death is nil:

$$\lim_{s \to +\infty} \tilde{R}_{t,s} [B_{a+s,t+s} + B^*_{a+s,t+s}] = 0$$
(10)

where  $\tilde{R}_{t,s}$  is a subjective discount factor applied by the household to its income streams, which incorporates the probability of survival, implying overdiscounting of events further apart in time. This is defined as:

$$\tilde{R}_{t,s} = \begin{cases} 1 \quad \text{for} \quad s = 0 \\ \\ \prod_{l=1}^{s} \frac{\theta}{i_{t+l-1}} \quad \text{for} \quad s > 0 \end{cases}$$
(11)

Secondly, we write current consumption of cohort a, in nominal terms, as a function of financial and human wealth:

$$P_t C_{a,t}^{OLG} \Theta_t = H W_{a,t} + F W_{a,t} \tag{12}$$

where  $\Theta_t$  is the inverse of  $mpc_t$ , and is given by:

$$\Theta_{t} = \frac{1}{\eta^{OLG}} + \frac{\theta \left(\frac{C_{t}^{OLG}}{C_{t-1}^{OLG}}\right)^{v\eta^{OLG}(1-\frac{1}{\gamma})} \left(\frac{w_{t+1\chi(1-\tau_{L,t+1})}}{w_{t}(1-\tau_{L,t})}\right)^{(1-\eta^{OLG})(1-\frac{1}{\gamma})} (r_{t}\beta)^{\frac{1}{\gamma}}}{r_{t}} \Theta_{t+1}$$
(13)

and  $HW_{a,t}$  corresponds to cohort *a*'s households' nominal human wealth, defined as the expected PDV of their labour supply endowments,  $HW_{a,t}^L$ , plus the expected PDV of their capital and dividend income net of lump-sum transfers/payments to the government,  $HW_{a,t}^K$ . Aggregating these variables across cohorts, we obtain:

$$hw_t = hw_t^L + hw_t^K \tag{14}$$

$$hw_t^L = n(1-\psi)w_t(1-\tau_{L,t}) + \frac{\theta\chi}{r_t}hw_{t+1}^L$$
(15)

$$hw_t^K = (1-\iota) \sum_{D=N,T,C,G,I,X} d_t^D + (1-\psi) \left( d_t^U + trg_t + trx_t \right) + \frac{\theta}{r_t} hw_{t+1}^K$$
(16)

where all variables are at their aggregate, real levels and  $\iota$  is the fraction of firms' dividends that are transferred from OLG to LIQ households.

 $FW_{a,t}$ , in turn, is the financial wealth stock, composed of households' domestic and foreign current liabilities. Aggregating across cohorts we obtain:

$$fw_t = r_{t-1} \left[ b_{t-1} + b_{t-1}^* \epsilon_{t-1} \right]$$
(17)

where again all variables are expressed in their aggregate, real levels and  $\epsilon_t = \frac{P_t^*}{P_t} = p_t^*$  is the real exchange rate. It should be noted that we have adopted the convention that each nominal asset

is deflated by the consumption based price index of the issuer, so that real domestic bonds are  $b_t = \frac{B_t}{P_t}$  and real foreign bonds are  $b_t^* = \frac{B_t^*}{P_t^*}$ .

Finally, aggregating equation (12) over the different cohorts, the relation between consumption and wealth can then be written as:

$$C_t^{OLG} = \frac{1}{\Theta_t} \left( hw_t + fw_t \right) \tag{18}$$

The above equation highlights an important feature of OLG households. The government services its liabilities through different forms of taxation. These future taxes affect different components of the human wealth as well as the marginal propensity to consume out of wealth. But unlike the government, which is infinitely lived, households take into account that they may not live enough to be responsible for higher future tax payments. Hence, each household discounts future tax liabilities by a rate of at least  $\frac{r_t}{\theta}$ , which is higher than the market's real interest rate  $r_t$ , as reflected in the discount factors in (13), (15) and (16). The household's discount rate for the labour income component of human wealth is even higher,  $\frac{r_t}{\theta\chi}$ , due to the role played by the decline of labour income over the household's life cycle. The implication from the household's attach a positive probability to the event of not being responsible for paying the taxes necessary to service that debt. The higher the instant death rate, the greater the portion of outstanding government debt households will consider to be net wealth.

#### 3.1.2 LIQ households

Analogously to the OLG case, a LIQ household h with age a rents labour services to a union, receiving  $W_t \Phi_a$ , and pays a labour income tax to the government at the tax rate  $\tau_{L,t}$ . Furthermore, it also receives transfers from the government,  $TRH_{a,t} + TRG_{a,t}^{LIQ}$ , from abroad,  $TRX_{a,t}^{LIQ}$ , and a share in labour union dividends, corresponding to the wage wedge created by the fact that its labour service is differentiated.

The LIQ household's budget constraint in nominal terms is thus given by:

$$P_t C_{a,t}^{LIQ}(h) = W_t \Phi_{a,t} L_{a,t}^{LIQ}(h) (1 - \tau_{L,t}) + TRH_{a,t}(h) + TRG_{a,t}^{LIQ}(h) + TRX_{a,t}^{LIQ}(h)$$
(19)

The LIQ household's optimisation problem is then reduced to the choice of consumption and labour supply that maximise the PDV of her future utility stream, subject to the constraints imposed by (1), (3) and (19). Analogously to OLG households we obtain the following expression for the aggregate consumption/leisure optimal allocation:

$$\frac{C_t^{LIQ}}{n\psi - L_t^{LIQ}} = \frac{\eta^{LIQ}}{1 - \eta^{LIQ}} w_t (1 - \tau_{L,t})$$
(20)

Furthermore, aggregating the budget constraint, we get the following expression for the real aggregate consumption of LIQ households:

$$C_t^{LIQ} = w_t L_t^{LIQ} (1 - \tau_{L,t}) + \iota \left( d_t^N + d_t^T + d_t^C + d_t^G + d_t^I + d_t^X \right) + \psi \left( d_t^U + trg_t + trx_t \right)$$
(21)

## 3.2 Labour unions

Labour unions hire the differentiated labour services from households and rent it to manufacturers at a higher wage than the one they pay. This way, they explore the market power stemming from labour differentiation, by creating a wedge between the wage received by households and the wage paid by manufacturers. The resulting monopoly rents are then distributed to households in the form of dividends, so that in practice they are the ones who benefit from the labour market power. Each union represents a particular variety of labour attached to a particular household, being therefore identified by the households' index h.

Each manufacturer j demands a homogeneous labour service,  $U_t(j)$ , composed of different varieties of labour,  $U_t(h, j)$ , which are bundled up using the following CES technology:

$$U_t(j) = \left(\int_0^1 U_t(h,j)^{\frac{\sigma_{U,t}-1}{\sigma_{U,t}}} dh\right)^{\frac{\sigma_{U,t}}{\sigma_{U,t}-1}}$$
(22)

where  $0 \leq \sigma_{U,t} \leq \infty$  is the elasticity of substitution across different varieties of labour, modelled as a shock.<sup>15</sup>

Manufacturers minimise the cost of obtaining the different varieties of labour, subject to the quantity of homogeneous labour services they demand, given in (22), which produces the following demand for each variety h, by a representative manufacturer j:

$$U_t(h,j) = \left(\frac{V_t(h)}{V_t}\right)^{-\sigma_{U,t}} U_t(j)$$
(23)

where  $V_t(h)$  is the wage charged by union h and  $V_t$  is the aggregate wage level.

<sup>&</sup>lt;sup>15</sup>In the context of the current study  $\sigma_{U,t}$  plays a prominent role, since it uniquely determines the degree of monopoly in the labour market. In practice, the fiercer the competition, the higher the elasticity of substitution and the lower the labour unions' marker power.

Integrating (23) across manufacturers, we get the following aggregate demand for variety h:

$$U_t(h) = \left(\frac{V_t(h)}{V_t}\right)^{-\sigma_{U,t}} U_t \tag{24}$$

where  $U_t$  stands for the aggregate demand for labour services by manufacturers.

Using (23) and (22) it is straightforward to get the aggregate wage as a function of the wages charged for each variety of labour and the elasticity of substitution:

$$V_t = \left(\int_0^1 V_t(h)^{1-\sigma_{U,t}} dh\right)^{\frac{1}{1-\sigma_{U,t}}}$$
(25)

In order to feature sticky wage growth, we have imposed quadratic adjustment costs following Kim (2000) and Laxton and Pesenti (2003). The choice of quadratic wage growth adjustment costs instead of staggered wage adjustment suggested by Erceg, Henderson and Levin (2000) with indexation, was mainly due to the higher analytical tractability of the first, in the context of OLG models. Moreover, Rotemberg (1987) has shown that in practice the two schemes yield similar reduced form Phillips curves and therefore are not empirically distinguishable in aggregate terms. Wage growth adjustment costs are then given by:

$$\Gamma_t^U(h) = \frac{\phi_U}{2} T_t U_t \left( \frac{\frac{V_t(h)}{V_t - 1(h)}}{\frac{V_{t-1}}{V_{t-2}}} - 1 \right)^2$$
(26)

where  $\phi_U$  determines how costly it is to adjust wage growth rate.

The expected future dividend stream of labour union h is given by:

$$E_t \sum_{s=0}^{\infty} \tilde{R}_{t+s} (1 - \tau_{L,t}) \left[ V_t(h) U_t(h) - W_t U_t(h) - P_t \Gamma_t^U(h) \right]$$
(27)

The dividend in each period corresponds to the (after tax) union's surplus from selling labour services for a higher price than it paid,  $V_t(h)U_t(h) - W_tU_t(h)$ , minus the costs paid for adjusting wages,  $P_t\Gamma_t^U(h)$ . We considered that labour union's dividends are subject to the same tax rate as wage income, to ensure that households are indifferent between wages or labour union dividends. The intertemporal discount rate applied to each period's dividend is  $\tilde{R}_{t+s}$  so as to equate the discount factor of labour unions with the subjective discount factor for income streams of their owners, the households.

The labour union will operate like a firm, deciding on the optimal wage to charge to manufacturers,  $V_t(h)$ , so as to maximise the PDV of its future dividend stream, subject to (24) and (26). The FOC of this problem yields the wage Phillips curve and, as in the case of households, there can be assumed a symmetric equilibrium in which all unions will behave alike and, thus, household indexes h can be suppressed:

$$\frac{\sigma_{U,t}}{\sigma_{U,t}-1}w_t - v_t = \frac{\phi_U}{\sigma_{U,t}-1} \left[ T_t \left( \frac{\pi_t^V}{\pi_{t-1}^V} - 1 \right) \frac{\pi_t^V}{\pi_{t-1}^V} - \frac{1 - \tau_{L,t+1}}{1 - \tau_{L,t}} \frac{\theta}{r_t} \frac{U_{t+1}T_{t+1}}{U_t} \left( \frac{\pi_{t+1}^V}{\pi_t^V} - 1 \right) \frac{\pi_{t+1}^V}{\pi_t^V} \right]$$
(28)

where  $\pi_t^V = \frac{V_{t+1}}{V_t}$  stands for wage inflation. In the steady-state this Phillips curve simply states that the wage charged to manufacturers is a constant markup on the households' wage rate.

The wage markup, which is in the core of this study, reflects the ability of labour unions to explore the market power arising from labour differentiation. Being differentiated, labour services are not easily substitutable and therefore manufacturers will be willing to pay a higher wage than they would if labour was homogeneous. In practice, this implies that workers will receive more than their marginal disutility of working and can therefore choose to work a bit less and still receive the wage income that they would if they worked harder and were strictly compensated by their disutility of working. This is a friction, in the sense that it limits the quantity of labour supplied, yielding an outcome which is suboptimal from a social point of view. Therefore, reforms aimed at increasing competition in the labour market, and in this way reducing the markup wedge, are likely to promote higher employment levels and stimulate the economy as a whole.

#### 3.3 Firms

Firms are of two types: manufacturers and distributors. Manufacturers produce differentiated tradable and non-tradable intermediate goods, using capital and labour services as inputs. Intermediate goods are then sold to distributors who combine them with imported goods to produce four types of differentiated final goods: private consumption, investment, government consumption and export goods.

All firms in the economy operate in monopolistic competition in their output markets, charging a markup over their marginal cost, and in perfect competition in their input markets, rewarding production factors at their marginal productivity level.

To obtain realistic impulse responses, in particular a reasonable short-run behaviour, a number of nominal and real rigidities are featured in the model, namely in investment, prices and imports.

#### 3.3.1 Manufacturers

Manufacturers (indexed by  $J \in \{T, N\}$ ) produce intermediate goods of two types: tradable (T) and non-tradable (N) goods. For each type of intermediate good, there is a continuum of manufacturing firms (indexed by  $j \in (0, 1)$ ), each one producing a different variety of the good.

Each distributor f demands a homogeneous intermediate good of type J,  $Z_t^J(f)$ , composed of a certain amount of each variety of intermediate goods of type J,  $Z_t^J(j, f)$ , which are bundled up using a CES technology. Analysing this distributor-manufacturer relationship in a perfectly analogous way as we did with the manufacturer-labour union one, we obtain the following aggregate demand for variety j of type J intermediate good:

$$Z_t^J(j) = \left(\frac{P_t^J(j)}{P_t^J}\right)^{-\sigma_{J,t}} Z_t^J$$
(29)

where  $Z_t^J$  stands for the aggregate demand for intermediate good of type J,  $P_t^J(j)$  is the price charged for variety j,  $P_t^J$  is the price index of type J good and  $0 \le \sigma_{J,t} \le \infty$  is the elasticity of substitution between type J good varieties, modelled as a shock.

Furthermore, we obtain an expression for the aggregate price of intermediate good of type J as a function of the prices of all its varieties and the elasticity of substitution:

$$P_t^J = \left(\int_0^1 P_t^J(j)^{1-\sigma_{J,t}} dj\right)^{\frac{1}{1-\sigma_{J,t}}}$$
(30)

To produce its good, each firm uses a CES technology, combining labour services,  $U_t^J(j)$ , with capital stock,  $K_t^J(j)$ :

$$Z_{t}^{J}(j) = \left( (1 - \alpha_{U}^{J})^{\frac{1}{\xi_{ZJ}}} \left( K_{t}^{J}(j) \right)^{\frac{\xi_{ZJ} - 1}{\xi_{ZJ}}} + (\alpha_{U}^{J})^{\frac{1}{\xi_{ZJ}}} \left( T_{t} A^{J} U_{t}^{J}(j) \right)^{\frac{\xi_{ZJ} - 1}{\xi_{ZJ}}} \right)^{\frac{\xi_{ZJ} - 1}{\xi_{ZJ} - 1}}$$
(31)

where  $0 \leq \xi_{ZJ} \leq \infty$  is the sector-specific elasticity of substitution between capital and labour and  $0 \leq \alpha_U^J \leq 1$  is the CES distribution parameter.  $T_t$  is a labour-augmenting technical progress trend, while  $A^J$  is a sector-specific technology scaling factor. It should be noted that  $T_t$  is the only source of real growth in the steady-state. In practice, we consider that  $T_t$  evolves in a purely deterministic fashion such that  $T_t = gT_{t-1}$ , where  $g \geq 1$  is the steady-state quarterly growth rate.

As already mentioned, nominal and real adjustment costs are featured. These are assumed to be quadratic, following the same reasoning presented in the wages case. On the nominal side, price growth adjustment costs, which ensure reasonable price dynamics, are given by:

$$\Gamma_t^{PJ}(j) = \frac{\phi_{PJ}}{2} Z_t^J \left( \frac{\frac{P_t^J(j)}{P_{t-1}^J(j)}}{\frac{P_t^J}{P_{t-2}^J}} - 1 \right)^2$$
(32)

where  $\phi_{PJ}$  is the sector-specific cost adjustment parameter.

On the real side, investment adjustment costs, which ensure a smooth response of capital/investment to changes in the desired capital stock, are given by:

$$\Gamma_t^{IJ}(j) = \frac{\phi_{K^J}}{2} K_t^J(j) \left(\frac{I_t^J(j)}{K_t^J(j)} - \frac{I^J}{K^J}\right)^2 + \frac{\phi_{IJ}}{2} K_t^J(j) \left(\frac{I_t^J(j)}{K_t^J(j)} - \frac{I_{t-1}^J}{K_{t-1}^J}\right)^2$$
(33)

where  $\phi_{K^J}$  and  $\phi_{I^J}$  are sector-specific investment adjustment costs,  $\frac{I_t^J(j)}{K_t^J(j)}$  is the sector-specific investment-to-capital ratio in period t and  $\frac{I^J}{K^J}$  is the corresponding steady-state level.

A fixed cost term,  $T_t \omega^J$ , is also assumed to ensure that economic profits arising from monopolistic competition are depleted in the steady-state and that, therefore, there are no firms entering or leaving the market.<sup>16</sup>

As usual, the capital stock results from the accumulation of investment goods,  $I_t^J(j)$ , with the capital accumulation rule being given by:

$$K_{t+1}^{J}(j) = (1 - \delta^{J})K_{t}^{J}(j) + I_{t}^{J}(j)$$
(34)

where  $0 \leq \delta^J \leq 1$  stands for the sector-specific capital depreciation rate.

The expected dividend stream of manufacturer j in sector J,  $D_t^J(j)$ , is defined as follows:

$$E_{t} \sum_{s=0}^{\infty} \tilde{R}_{t+s} \left\{ P_{t}^{J}(j) Z_{t}^{J}(j) - (1+\tau_{SP}) V_{t} U_{t}^{J}(j) - P_{t}^{I} \left( I_{t}^{J}(j) + \Gamma_{t}^{IJ}(j) \right) - P_{t}^{J} \left( \Gamma_{t}^{PJ}(j) + T_{t} \omega^{J} \right) - \tau_{K} \left[ P_{t}^{J}(j) Z_{t}^{J}(j) - (1+\tau_{SP}) V_{t} U_{t}^{J}(j) - P_{t}^{I} \left( q_{t}^{J} \delta^{J} K_{t}^{J}(j) + \Gamma_{t}^{IJ}(j) \right) - P_{t}^{J} \left( \Gamma_{t}^{PJ}(j) + T_{t} \omega^{J} \right) \right] \right\}$$
(35)

The dividend in each period corresponds to the operational cashflow of the firm, less the capital income taxes paid to the government. As already explained in the unions' case, the discount rate applied to each period's dividend is  $\tilde{R}_{t+s}$ . The operational cashflow corresponds to overall revenue,  $P_t^J(j)Z_t^J(j)$ , minus overall expenditure, which includes the wage bill and social security contributions,  $(1 + \tau_{SP})V_tU_t^J(j)$ , investment expenditure,  $P_t^I I_t^J(j)$ , and adjustment and fixed cost expenditures,  $P_t^I \Gamma_t^{IJ}(j) + P_t^J \Gamma_t^{PJ}(j) + P_t^J T_t \omega^J$ . The capital income tax is applied only to operational net profits  $P_t^J(j)Z_t^J(j) - (1 + \tau_{SP})V_tU_t^J(j) - P_t^I q_t^J \delta^J K_t^J(j) - P_t^I \Gamma_t^{IJ}(j) - P_t^J \Gamma_t^{PJ}(j) - P_t^J T_t \omega^J$ , where  $q_t^J$  is Tobin's-Q, the shadow price of a unit of installed capital in terms of current investment goods, which enables a correct valuation of the capital stock at each point in time.<sup>17</sup>

The optimal plan of manufacturer j consists in setting labour demand, capital stock, investment and price, in such a way as to maximise the PDV of its future dividend stream subject to (31), (32), (33), (34) and (29). This yields FOCs for the intermediate good's price, labour demand, capital stock level and investment path. As in the households and unions' case, all manufacturing firms operate symmetrically, and therefore the aggregation is straightforward and the individual indexes j can be dropped from the optimality conditions. The optimal price-setting condition yields the

<sup>&</sup>lt;sup>16</sup>The fixed cost term is defined as a constant share of output, ensuring that it does not vanish along the balanced growth path of the economy as in Kumhof and Laxton (2007b).

<sup>&</sup>lt;sup>17</sup>For fiscal purposes, the relevant concept is net operational profits and not operational cashflow. While the cashflow records all expenditures on capital goods, profits only record the economic depreciation of the installed capital stock. This means, for instance, that an increase in investment implies a lower dividend but does not affect the tax due on operational profits.

following Phillips curve for the inflation rate of type J intermediate good,  $\pi_t^J = \frac{P_t^J}{P_{t-1}^J}$ :

$$\frac{\sigma_{J,t}}{\sigma_{J,t}-1}\frac{\lambda_t^J}{p_t^J} - 1 = \frac{\phi_{P^J}}{\sigma_{J,t}-1} \left[ \left( \frac{\pi_t^J}{\pi_{t-1}^J} - 1 \right) \frac{\pi_t^J}{\pi_{t-1}^J} - \frac{\theta}{r_t} \frac{p_{t+1}^J}{p_t^J} \frac{Z_{t+1}^J}{Z_t^J} \left( \frac{\pi_{t+1}^J}{\pi_t^J} - 1 \right) \frac{\pi_{t+1}^J}{\pi_t^J} \right]$$
(36)

where  $\lambda_t^J$  stands for the real marginal cost of type J intermediate good. In the steady-state, this Phillips curve collapses to a standard markup condition, implying that the price of intermediate good of type J is simply a markup over its marginal cost. As in the case of labour unions, the markup over the marginal cost charged by the manufacturer of type J good depends solely on  $\sigma_{J,t}$ , which is the degree of differentiation between varieties of the intermediate good at stake and reflects, among other factors, the institutional framework in the product market. Following the same reasoning as applied in the wages case, any reforms aimed at increasing competition level in the goods market are likely to decrease the price markup and to induce a higher level of output, thus affecting international competitiveness and income levels in non-negligible magnitudes.

The optimal labour demand condition is quite standard and equates the marginal productivity of labour to its (real) marginal cost, which includes employers' social contributions, besides the hourly wage charged by labour union, as follows:

$$(1+\tau_{SP})\frac{v_t}{\lambda_t^J} = \left(\frac{Z_t^J \alpha_U^J}{T_t A^J U_t^J}\right)^{\frac{1}{\xi_{ZJ}}} T_t A^J$$
(37)

The optimal capital condition is given by:

$$\frac{1}{1-\tau_{K}} \left( q_{t}^{J} \frac{r_{t}}{\theta} \frac{\pi_{t+1}}{\pi_{t+1}^{I}} - q_{t+1}^{J} \right) = \frac{I_{t+1}^{J}}{K_{t+1}^{J}} \left[ \phi_{K}^{J} \left( \frac{I_{t+1}^{J}}{K_{t+1}^{J}} - \frac{I^{J}}{K^{J}} \right) + \phi_{I}^{J} \left( \frac{I_{t+1}^{J}}{K_{t+1}^{J}} - \frac{I_{t}^{J}}{K_{t}^{J}} \right) \right] - \left[ \frac{\phi_{K}^{J}}{2} \left( \frac{I_{t+1}^{J}}{K_{t+1}^{J}} - \frac{I^{J}}{K^{J}} \right)^{2} + \frac{\phi_{I}^{J}}{2} \left( \frac{I_{t+1}^{J}}{K_{t+1}^{J}} - \frac{I_{t}^{J}}{K_{t}^{J}} \right)^{2} \right] + \frac{r_{K,t+1}^{J}}{p_{t+1}^{I}} - q_{t+1}^{J} \delta^{J}$$
(38)

The above equation is quite cumbersome and it involves all the elements related with the cost of capital (the depreciation rate, the real interest rate, Tobin's-Q and the price of capital goods), the real rate of return on capital,  $r_{K,t+1}^J$ , and the adjustment costs associated to investment. In the steady-state, this equation states that the capital stock must be such that the after-tax real return of investing in capital and adjusting for depreciation and capital goods' market value must equal the certain equivalent real return of investing in government bonds.

The investment demand equation relates Tobin's-Q with investment adjustment costs and the capital income tax:

$$q_t^J = 1 + (1 - \tau_K) \left[ \phi_K^J \left( \frac{I_t^J}{K_t^J} - \frac{I^J}{K^J} \right) + \phi_I^J \left( \frac{I_t^J}{K_t^J} - \frac{I_{t-1}^J}{K_{t-1}^J} \right) \right]$$
(39)

The above condition basically implies that the furthest the Tobin's-Q is from unity, the more prone the investor is to incur in investment adjustment costs. Since Tobin's-Q is the shadow value of a unit of installed capital in terms of the market value of capital goods, then a value above (below) unity means that the market value of capital goods is relatively cheaper (more expensive) and thus capital stock must rise (decline), implying that investment to capital ratio must temporarily increase (decrease). In the steady-state, Tobin's-Q is equal to one, meaning that the shadow value of installed capital equals its market value, exhausting arbitrage opportunities.

#### 3.3.2 Distributors

Distributors (indexed by  $F \in \{C, G, I, X\}$ ) produce four types of final goods: consumption goods (C), investment goods (I), government consumption goods (G) and export goods (X). For each type of final good, there is a continuum of distributors (indexed by  $f \in (0, 1)$ ), each one producing a different variety of the good. The four types of distributors use a similar technology, which involves a two stage production process described further on in this section. Similarly to manufacturers, final goods distributors operate in perfect competition in their input markets and in monopolistic competition in their output markets, taking advantage of the monopoly power created by final goods differentiation, charging a markup on marginal costs.

Each type of final good is demanded by a unique type of costumer: private consumption goods are demanded by households, government consumption goods are demanded by the government, capital goods are demanded by manufacturing firms and export goods are demanded by foreign agents. For each type of costumer (indexed by  $E \in \{C, G, I, X\}$ ) there is a continuum of agents (indexed by  $e \in (0,1)$ ), who demand a certain amount of final good of type F,  $Y_t^F(e)$ , obtained by bundling up the different varieties of final good,  $Y_t^F(f, e)$ . This costumer-distributer relation is perfectly analogous to what has been presented for labour unions and manufacturers, and produces the following demand for variety f of type F intermediate good:

$$Y_t^F(f) = \left(\frac{P_t^F(f)}{P_t^F}\right)^{-\sigma_F} Y_t^F \tag{40}$$

where  $Y_t^F$  stands for the aggregate demand for type F final good, and  $P_t^F(f)$  is the price charged by the distributor for variety f,  $P_t^F$  is the aggregate price index of type F good and  $0 \le \sigma_F \le \infty$ is the elasticity of substitution between different varieties of type F good.

We also obtain the following expression for the aggregate price index of type F final good:

$$P_t^F = \left(\int_0^1 P_t^F(f)^{1-\sigma_F} df\right)^{\frac{1}{1-\sigma_F}}$$
(41)

The two stages of the distribution production process, named the assemblage stage and the

distribution stage, will now be described.

#### Stage 1: the assemblage stage

In this stage, the distributor combines the domestic tradable good,  $Z_t^{TF}(f)$ , with the imported good,  $M_t^F(f)$ , through a CES technology, to obtain a composite differentiated good,  $Y_t^{AF}(f)$ , henceforth called the assembled good.

To obtain a realistic behaviour of the import contents, whenever the economy is hit by shocks that imply significant real exchange rate fluctuations, we consider import content adjustment costs with the following specification:

$$\Gamma_t^{AF}(f) = \frac{\phi_{AF}}{2} \frac{\left(\mathcal{A}_t^{AF}(f) - 1\right)^2}{1 + \left(\mathcal{A}_t^{AF}(f) - 1\right)^2} \quad \text{with} \quad \mathcal{A}_t^{AF}(f) = \frac{M_t^F(f)/Y_t^{AF}(f)}{M_{t-1}^F/Y_{t-1}^{AF}} \tag{42}$$

where  $\phi_{AF}$  is the sector-specific parameter determining the adjustment cost of the assembled good import content.

The production function for variety f of the assembled good of type F is then given by:

$$Y_{t}^{AF}(f) = \left( (\alpha_{AF})^{\frac{1}{\xi_{AF}}} \left( Z_{t}^{TF}(f) \right)^{\frac{\xi_{AF}-1}{\xi_{AF}}} + (1 - \alpha_{AF})^{\frac{1}{\xi_{AF}}} \left( M_{t}^{F}(f) \left[ 1 - \Gamma_{t}^{AF}(f) \right] \right)^{\frac{\xi_{AF}-1}{\xi_{AF}}} \right)^{\frac{\xi_{AF}-1}{\xi_{AF}}}$$
(43)

where  $0 \leq \xi_{AF} \leq \infty$  is the elasticity of substitution between domestic tradable good and the imported good and  $0 \leq \alpha_{AF} \leq 1$  is the home bias parameter.

The optimisation problem of a distributor of variety f of type F final good at this stage consists in setting the demand for both the tradable domestic good and the imported good so as to minimise the cost of producing the desired quantity of assembled good, subject to (43) and (42). The FOCs of this problem set optimal demand for production factors. Since all distributors of type F final good solve the same cost minimisation problem under the same conditions, we can assume a symmetric equilibrium and drop the individual identifier f. The demand for domestic tradable good by type F final good distributors is then given by:

$$Z_t^{TF} = \alpha_{AF} \left(\frac{p_t^T}{\lambda_t^{AF}}\right)^{-\xi_{AF}} Y_t^{AF} \tag{44}$$

where  $\lambda_t^{AF}$  stands for the marginal cost of production of the type F assembled good. This demand function simply states that the desired quantity of tradable good depends on the amount of type Fassembled good being produced, its domestic tradable goods content, its price in terms of the firm's marginal cost  $\frac{p_t^T}{\lambda_t^{AF}}$  and on the elasticity of substitution between domestic tradable and imported goods. The demand for imported good by type F final good distributors is in turn given by:

$$M_{t}^{F} = (1 - \alpha_{AF}) \left(\frac{p_{t}^{*}}{\chi_{t}^{AF} \lambda_{t}^{AF}}\right)^{-\xi_{AF}} \frac{Y_{t}^{AF}}{1 - \Gamma_{t}^{AF}} \quad \text{with} \quad \chi_{t}^{AF} = 1 - \Gamma_{t}^{AF} - \phi_{FT} \frac{(\mathcal{A}_{t}^{AF} - 1)\mathcal{A}_{t}^{AF}}{\left(1 + (\mathcal{A}_{t}^{AF} - 1)^{2}\right)^{2}}$$
(45)

where  $P_t^*$  stands for the price of the imported good, which is in any case set in euros, since the euro area plays the role of the foreign economy. This equation is very similar to equation (44) and its interpretation is identical in the steady-state, when  $\chi_t^{AF} = 1$  and  $\Gamma_t^{AF} = 0$ . However, in the short run, when import contents are changing, adjustment costs arise with  $\chi_t^{AF} > 1$  and  $\Gamma_t^{AF} > 0$ , implying that *ceteris paribus* more imported good needs to be used to obtain the same amount of assembled final good, due to the inefficiencies in the adjustment of the import content.

Using (43), (44) and (45),  $\lambda_t^{AF}$  can easily be derived:

$$\lambda_t^{AF} = \left[ \alpha_{AF} (p_t^T)^{1 - \xi_{AF}} + (1 - \alpha_{AF}) \left( \frac{p_t^*}{\chi_t^{AF}} \right)^{1 - \xi_{AF}} \right]^{\frac{1}{1 - \xi_{AF}}}$$
(46)

#### Stage 2: the distribution stage

In this stage, the distributor combines the assembled good previously produced,  $Y_t^{AF}(f)$ , with the domestic non-tradable good,  $Z_t^{NF}(f)$ , to obtain the final output,  $Y_t^F(f)$ . This is achieved through the following CES technology:

$$Y_t^F(f) = \left( (1 - \alpha_F)^{\frac{1}{\xi_F}} \left( Y_t^{AF}(f) \right)^{\frac{\xi_F - 1}{\xi_F}} + (\alpha_F)^{\frac{1}{\xi_F}} \left( Z_t^{NF}(f) \right)^{\frac{\xi_F - 1}{\xi_F}} \right)^{\frac{\xi_F}{\xi_F - 1}}$$
(47)

where  $0 \leq \xi_F \leq \infty$  is the elasticity of substitution between the assembled good and the non-tradable good and  $0 \leq \alpha_F \leq 1$  is the non-tradable good distribution parameter.

At this stage, distributors are subject to price growth adjustment costs, which follow a quadratic form identical to the one assumed for manufacturers:

$$\Gamma_t^{PF}(f) = \frac{\phi_{PF}}{2} Y_t^F \left( \frac{P_t^F(f) / P_{t-1}^F(f)}{P_{t-1}^F / P_{t-2}^F} - 1 \right)^2$$
(48)

where  $\phi_{PF}$  is the sector-specific price adjustment cost parameter for firms operating in sector Fand  $P_t^F(f)$  is the price charged by the distributor for their output.

Analogously to manufacturers, a fixed cost is assumed,  $T_t \omega^F$ , which assures that economic profits arising from monopolistic competition are depleted in the steady-state.

The expected dividend income stream for distributor f of type F good can be represented as:

$$E_{t} \sum_{s=0}^{\infty} \tilde{R}_{t+s} \left(1 - \tau_{D}\right) \left[ P_{t}^{F}(f) \left( Y_{t}^{F}(f) - \Gamma_{t}^{PF}(f) - T_{t} \omega^{F} \right) - \Lambda_{t}^{AF}(f) Y_{t}^{AF}(f) - P_{t}^{N} Z_{t}^{NF}(f) \right]$$
(49)

which corresponds to the after-tax difference between revenue,  $P_t^F(f)Y_t^F(f)$ , and expenditure, which includes intermediate consumption costs,  $\Lambda_t^{AF}(f)Y_t^{AF}(f) + P_t^N Z_t^{NF}(f)$ , and the inefficiencies imposed by adjustment and fixed costs,  $P_t^F(f) \left(\Gamma_t^{PF}(f) + T_t \omega^F\right)$ . As in the case of labour unions and manufacturing firms, the intertemporal discount rate applied to each period's dividend is  $\tilde{R}_{t+s}$ .

The optimal plan of distributor f of type F consists in setting the demand for assembled good and non-tradable good, and the price of final good output so as to maximise the PDV of future dividend stream, subject to constraints imposed by (47), (48) and (40). The solution of this problem is a set of optimality conditions for input demand and price setting. Again, since all distributors are similar, symmetric equilibrium can be imposed and aggregate conditions can be obtained and f can be suppressed. The demand for assembled final good of type F, is given by:

$$Y_t^{AF} = (1 - \alpha_F) \left(\frac{\lambda_t^{AF}}{\lambda_t^F}\right)^{-\xi_F} Y_t^F$$
(50)

and, similarly, the demand condition for non-tradable good is given by:

$$Z_t^{NF} = \alpha_F \left(\frac{p_t^N}{\lambda_t^F}\right)^{-\xi_F} Y_t^F \tag{51}$$

Using (50) and (51) in (47),  $\lambda_t^F$  can easily be derived:

$$\lambda_t^F = \left[ (1 - \alpha_F) (\lambda_t^{AF})^{1 - \xi_F} + \alpha_F (p_t^N)^{1 - \xi_F} \right]^{\frac{1}{1 - \xi_F}}$$
(52)

The optimal price-setting condition yields the following Phillips curve for the inflation rate of type F final good,  $\pi_t^F = \frac{P_t^F}{P_{t-1}^F}$ :

$$\frac{\sigma_F}{\sigma_F - 1} \frac{\lambda_t^F}{p_t^F} - 1 = \frac{\phi_{PF}}{\sigma_F - 1} \left[ \left( \frac{\pi_t^F}{\pi_{t-1}^F} - 1 \right) \frac{\pi_t^F}{\pi_{t-1}^F} - \frac{\theta}{r_t} \frac{p_{t+1}^F}{p_t^F} \frac{Y_{t+1}^F}{Y_t^F} \left( \frac{\pi_{t+1}^F}{\pi_t^F} - 1 \right) \frac{\pi_{t+1}^F}{\pi_t^F} \right]$$
(53)

In the steady-state, this condition has the usual Phillips curve interpretation.

#### 3.4 Government

In this model, the government has two main activities: consuming and performing transfers across households through taxes and subsidies. To finance its activities, the government levies taxes on labour income, firms' dividends and households' consumption, and benefits from non-tax revenues stemming from EU transfers. Government consumption operates as a pure distortion. Nobody benefits from it, since it does not enter in households' utility and has no impact on technology. In fact, the only thing it does is generate demand for a particular type of final good, which is extremely intensive in non-tradable intermediate goods and has a low import content. One must admit that this is a tremendous simplification of the government's role in the economy, since it fully neglects its role as an employer and investor and the impact of public services on households' utility. In addition, the model does not feature unemployment benefits, since it is assumed that all markets clear in each and every period, including the labour market. However, these simplifications are crucial to keep the model tractable from an analytical perspective.

In view of these simplifying assumptions, government consumption,  $G_t$ , and lump-sum government transfers to households,  $TRG_t$ , are fully exogenous and follow the exogenous real and nominal trends.<sup>18</sup> Besides these transfers, the government also performs transfers to LIQ households,  $TRH_t$ . These are fully funded within each period by lump-sum transfers charged to OLG households, and therefore do not affect per-period government budget constraint, being irrelevant from a fiscal perspective. The point of these transfers is simply to perform some redistribution from OLG to LIQ households so as to compensate the latter for not receiving dividend income.

On the revenue side, taxes are the most important source of revenue. Among them, *ad-valorem* consumption tax is crucial not only due to the share of its receipts in overall revenue, but also due to the distortion it generates, by creating a wedge between the price relevant to the consumption goods' distributor,  $P_t^C$ , and the price charged to households  $P_t$ . The revenue that accrues to the government from this form of taxation is:

$$RV_{C,t} = \tau_C P_t^C C_t \tag{54}$$

Labour income is taxed at the firm level, since manufacturers pay social security contributions to the government on their payroll, and at the household level, as households pay their labour income taxes. The revenue generated by taxes on labour income is then given by:

$$RV_{SP,t} + RV_{L,t} = \tau_{SP}V_tU_t + \tau_{L,t}\left(V_tU_t - P_t\Gamma_t^U\right)$$
(55)

The revenues from dividend income taxes paid by manufacturers and distributors can be represented similarly as a function of an average tax rate on operational profits as follows:

$$RV_{K,t} = \sum_{J=T,N} \tau_K \left[ P_t^J \left( Z_t^J - \Gamma_t^{PJ} - T_t \omega^J \right) - (1 + \tau_{SP}) V_t U_t^J - P_t^I \left( q_t^J \delta^J K_t^J - \Gamma_t^{IJ} \right) \right]$$
(56)

$$RV_{D,t} = \sum_{F=C,I,G,X} \tau_D \left[ P_t^F \left( Y_t^F - \Gamma_t^{PF} - T_t \omega^F \right) - P_t^T Z_t^{TF} - P_t^* M_t^F - P_t^N Z_t^{NF} \right]$$
(57)

In general, all tax rates but labour income tax are assumed to be exogenous. EU transfers,  $TRE_t$ , are also exogenously set and follow a nominal growth trend.

<sup>&</sup>lt;sup>18</sup>In practice, it corresponds to assume an exogenous government consumption to real output ratio and an exogenous transfers to private consumption ratio along the economy's balance growth path.

To finance its fiscal deficit, the government issues one-period government bonds,  $B_t$ . The government pays an interest rate,  $i_{t-1}$ , at the beginning of period t on the stock of government bonds held from period t - 1. For the sake of simplicity, full home bias in domestic government bond holding it assumed, meaning that foreigners do not hold this type of bond in their portfolios. This, however, is not a very stringent assumption, since OLG households may borrow from abroad to buy domestic government bonds. The government budget constraint can be represented as follows:

$$B_{t} = i_{t-1}B_{t-1} - \sum_{A=C,L,SP,K,D} RV_{A,t} - TRE_{t} + P_{t}^{G}G_{t} + TRG_{t}$$
(58)

To prevent an explosive debt path, a fiscal rule is imposed. The rule implies that at least one fiscal instrument (a variable fully controlled by the government) must adjust endogenously, ensuring that the fiscal surplus-to-GDP ratio,  $\left(\frac{SG}{GDP}\right)_t$ , converges to its target value,  $\left(\frac{SG}{GDP}\right)_t^{target}$ , which is univocally related to the steady-state debt-to-GDP ratio,  $\left(\frac{B}{GDP}\right)_t^{target}$ . As in Harrison et al. (2005), Kumhof and Laxton (2007b), Kumhof and Laxton (2007a) and Kilponen and Ripatti (2006), we consider that the government uses the labour income tax as the only fiscal instrument. We consider the structural balance fiscal rule proposed in Kumhof and Laxton (2008):

$$\left(\frac{SG}{GDP}\right)_{t} = \left(\frac{SG}{GDP}\right)_{t}^{target} + d_{tax}\left(\frac{RV_{t} - RV_{t}^{ss}}{GDP_{t}^{ss}}\right) + d_{debt}\left(\frac{B_{t}}{GDP_{t}^{ss}} - \left(\frac{B}{GDP}\right)_{t}^{target}\right)$$
(59)

where  $RV_t^{ss}$  is the tax revenue evaluated in the steady-state, that is the current average tax rates multiplied by tax bases at their respective steady-state levels and  $GDP_t$  and  $GDP_t^{ss}$  are observed and steady-state levels of nominal GDP, which will be defined below.

The fiscal policy rule adopted takes into consideration not only government debt stabilisation, but also business cycle smoothing. In terms of debt stabilisation, the rule adjusts automatically the labour income tax rate in order to generate sufficient revenue, preventing an explosive debt path. Since in the steady state  $RV_t = RV_t^{ss}$  and  $\frac{B_t}{GDP_t^{ss}} = \left(\frac{B}{GDP}\right)_t^{target}$ , the rule implies that  $\left(\frac{SG}{GDP}\right)_t^{target}$ converges to  $\left(\frac{SG}{GDP}\right)_t^{target}$ . As for business cycle smoothing, the rule implies a response of fiscal surplus-to-GDP ratio in each period that clearly depends on the nature of the shock affecting the economy. For instance, for  $d_{tax} = 1$ , the rule becomes a purely structural balance fiscal rule, the automatic stabilisers will operate at full extent, since realised fiscal surplus will rise with cyclical excess tax revenue. During a boom, for example, when the tax revenue is above its steady-state level, the government will pay off government debt with the extra funds, increasing fiscal surplus above its steady-state level. This structural fiscal balance rule minimises the variability of the labour tax rate and reduces the variability of output and inflation throughout the cycle relative to a balanced budget rule, obtained with  $d_{tax} = 0$ . A  $d_{tax} > 1$  implies a counter-cyclical fiscal policy rule. Finally, the  $d_{debt}$  parameter reflects the degree of government's aversion to deviations of the debt-to-GDP ratio from its target value.

#### 3.5 Rest of the world

The rest of the world (RW) is assumed to correspond to the members of the monetary union (excluding the domestic economy), which enables a simplification of the model, with the nominal effective exchange rate being irrevocably set to unity and all flows recorded in the same currency. The RW affects the home economy through trade and financial flows.

Regarding financial flows, since the home economy is small, changes in its net foreign assets do not affect the foreign interest rate.

As for as trade is concerned, we have already modelled imports' demand by domestic distributors and since we assume foreign prices to be exogenous, imports are fully determined. As for exports, while in a multi-country model these are endogenously determined, depending on comparative advantages among countries, in a single country small open economy model, their modelling is always an issue since the RW is not explicitly modelled. We follow the strategy used in Adolfson et al. (2005), with the home economy being assumed to be sufficiently small in relation to the RW economy, such that the impact of domestic shocks in the foreign economy is marginal enough to be fully neglected.

In this context, we derive the demand for domestic exports assuming that foreign distributors (indexed by  $f^* \in (0,1)$ ) demand an assembled good,  $Y_t^{A*}(f^*)$ , which results from an assembling of the domestic exported good,  $X_t(f^*)$ , with an intermediate tradable good,  $Z_t^{T*}(f^*)$  produced by foreign tradable goods manufacturers using the following CES technology:

$$Y_t^{A*}(f^*) = \left( (1 - \alpha^*)^{\frac{1}{\xi^*}} \left( Z_t^{T*}(f^*) \right)^{\frac{\xi^* - 1}{\xi^*}} + (\alpha^*)^{\frac{1}{\xi^*}} \left( X_t(f^*) \right)^{\frac{\xi^* - 1}{\xi^*}} \right)^{\frac{\xi^*}{\xi^* - 1}}$$
(60)

where  $\xi^*$  is the elasticity of substitution between foreign tradable goods and home exports and  $\alpha^*$  is the foreign economy bias parameter.

Each foreign distributor will set the demand for home country exports and tradable goods that minimise the cost of producing the desired quantity of assembled good, subject to the technology constraint imposed by (60). Since all foreign distributors solve the same cost minimisation problem under the same conditions, their optimal decisions will be identical and therefore we can assume symmetric equilibrium, dropping the  $f^*$  indexes. The demand for home exports is then given by the following standard demand curve equation:

$$X_t = \alpha^* \left(\frac{p_t^X}{p_t^{T*}}\right)^{-\xi^*} Y_t^{A*}$$
(61)

## 3.6 Market clearing conditions, final output and GDP

The model is closed by a set of market clearing conditions, which ensure that all markets clear in each and every period. In terms of the labour market, imposing that that overall labour supply by OLG and LIQ households must equal overall labour demand by manufacturers as follows:

$$L_t^{OLG} + L_t^{LIQ} = U_t^T + U_t^N \tag{62}$$

In the intermediate goods' market, the output produced by manufacturers in both sectors must meet demand by distributors and cover price adjustment costs and fixed costs as follows:

$$Z_t^T = Z_t^{TC} + Z_t^{TI} + Z_t^{TG} + Z_t^{TX} + \Gamma_t^{PT} + T_t\omega^T$$
(63)

$$Z_t^N = Z_t^{NC} + Z_t^{NI} + Z_t^{NG} + Z_t^{NX} + \Gamma_t^{PN} + T_t\omega^N$$
(64)

As for the final goods' market, the output supplied by each type of distributor must meet demand, adjustment costs and fixed costs, as follows:

$$Y_t^C = C_t^{OLG} + C_t^{LIQ} + \Gamma_t^{PC} + T_t \omega^C$$
(65)

$$Y_t^I = I_t^T + I_t^N + \Gamma_t^{TI} + \Gamma_t^{NI} + \Gamma_t^{PI} + T_t \omega^I$$
(66)

$$Y_t^G = G_t + \Gamma_t^{PG} + T_t \omega^G \tag{67}$$

$$Y_t^X = X_t + \Gamma_t^{PX} + T_t \omega^X \tag{68}$$

Finally, the net foreign asset accumulation closes the external block of the model, imposing that domestic savings must equal the change in foreign bond holdings:

$$B_t^* = i_{t-1}B_{t-1}^* + P_t^X X_t - P_t^* M_t + TRE_t + TRX_t$$
(69)

In addition, at this stage, it is useful to define GDP (at market prices) in line with the National Accounts (NA) and to distinguish it from final output (at market prices). Final output can be defined as the aggregate nominal value of the final goods made available by distributors:

$$P_t Y_t = P_t Y_t^C + P_t^I Y_t^I + P_t^G Y_t^G + P_t^X Y_t^X$$
(70)

This definition does not correspond to the NA definition of final demand, since it includes distributors' price adjustment and fixed costs. The NA nominal GDP is defined as the sum of the final expenditure of all agents in the economy, which corresponds to final demand, excluding the final expenditure that falls on imported goods,  $P_t^*M_t$ :

$$GDP_t = P_t C_t + P_t^G G_t + P_t^I I_t + P_t^X X_t - P_t^* M_t$$

$$\tag{71}$$

where overall imports,  $M_t$ , correspond to the sum of imports demand by all distributors,  $M_t^C + M_t^I + M_t^G + M_t^X$ . To compute chain-linked real GDP, one would have to calculate GDP at the previous year average price level and then derive the implied volume and deflator. However, in the context of this model, the relevant definition of real GDP is simply nominal GDP expressed in terms of the numeraire price.

#### 3.7 Shocks

There are three shocks in the model: one wage markup shock and two sector specific intermediate goods prices' markup shocks. These shocks follows a first order auto-regressive process with zero mean *iid* innovations and unconditional mean equal to their steady-state value.

$$\xi_t = \rho_{\xi}\xi_{t-1} + \eta_{\xi,t} \qquad \eta_{\xi,t} \sim N(0, \sigma_{\xi}^2) \tag{72}$$

where  $\xi = \{\sigma_T, \sigma_{NT}, \sigma_U\}.$ 

## 4 Calibrating *PESSOA* for the Portuguese economy

The calibration relied on data from the Portuguese economy and on other studies in the field. The data was taken from the Annual NA dataset available from the Portuguese statistics office and was mainly used to fix the steady-state parameters. The remaining parameters were calibrated using standard values in the DSGE literature and estimates for Portugal, whenever they are available.

The calibration figures for the model's main parameters are presented in Table 1. Being a small open economy in the euro area, Portugal's steady-state real GDP growth and inflation rate were assumed to be identical to those prevailing in the rest of the euro area, to ensure the existence of a balanced growth path. Therefore, labour-augmenting productivity's annual growth rate was set to 2%, which is consistent with the estimates for the euro area's potential output growth presented in Musso and Westermann (2005) and Proietti and Musso (2007) and also seems a plausible estimate for Portugal in view of the results of Almeida and Félix (2006). Following Coenen et al. (2007), the euro area nominal interest rate in the steady-state was set to 4.5%, whereas inflation was set to 2% in line with the ECB price stability definition. The elasticity of substitution between foreign tradables and Portuguese exports was set to 1.5, broadly in line with the estimate of Coenen et al. (2007), and the Portuguese goods content of foreign goods was calibrated to match the Portuguese economy's average exports-to-GDP ratio in the recent past, while foreign transfers to households were calibrated to broadly match the private transfers-to-GDP ratio in the balance of payments statistics (close to 2%).

The calibration of households' parameters took into consideration that the model comprehends Blanchard-Yaari overlapping generations, while most DSGE models consider the infinitely lived agents framework. These parameters were therefore largely based on Fagan, Gaspar and Pereira (2004), Harrison et al. (2005), Kumhof and Laxton (2007*a*) and Kumhof and Laxton (2007*b*).  $\eta_{OLG}$  and  $\eta_{LIQ}$  were calibrated so as to ensure that the elasticity of labour supply to real wage is 0.5, a value commonly found in the literature. Since the Blanchard-Yaari overlapping generations households framework allows for an endogenous determination of the net foreign asset position, the discount rate was calibrated to ensure a net foreign debt position of 60% in the steady-state. The coefficient of relative risk aversion was set to calibrate the intertemporal elasticity of substitution to 0.2, which might seem a low figure in comparison with the values typically used in infinitely lived agents models, but it is in the range of the values regularly used in models featuring Blanchard-Yaari households. The share of liquidity constrained households was set to 40%, broadly in line with the estimates for Portugal presented in Castro (2006).

In terms of labour unions' parameters, we considered a 25% wage markup, which is at the upper limit of the values usually found in the literature. Note, however, that since the labour market in Portugal is strongly regulated, one may argue that the markup could be even higher than the figures usually found in the DSGE literature. Nominal wage rigidity was calibrated to ensure that wages adjust to the new equilibrium in 6 quarters, a value slightly above euro area estimates published in Coenen et al. (2007), but still in the range usually found in the literature.

Turning to manufacturers, the depreciation rate was assumed to be identical across firms and was calibrated to get the investment-to-GDP ratio in line with the NA data. As regards the production function, a standard Cobb-Douglas function between capital and labour was assumed and the distribution parameters were calibrated to match the labour income share in the NA data. The price markup of tradable and non-tradable goods' manufacturers was calibrated using OECD product market regulation indicators and the correlation between tradable and non-tradable goods markups and product market regulation indicators found in Høj et al. (2007). In particular, the non-tradable goods markup was set to 20%, which is at the upper bound of the range of values commonly found in the literature, but consistent with the evidence pointing to low competition in the Portuguese non-tradable goods market that is at the core of the motivation for this study (see Section 2). As for real rigidities, capital adjustment costs were calibrated so as to ensure plausible impulse responses in terms of investment volatility. Regarding nominal rigidities, price growth adjustment costs were calibrated to match average adjustment time spans, in line with what is suggested in the literature. In particular, we impose that the adjustment of prices in the non-tradable goods' sector is slightly slower than in the tradable goods' sector, reflecting the fact that fiercer competition and lower markups imply lower price stickiness.

We now consider distributors' parameters. In the assemblage stage, the elasticity of substitution between domestic tradable goods and imports was taken to be identical across distributors and set above unity, as in most of the literature on open economy DSGE models (see for instance Coenen et al. (2007), Harrison et al. (2005), Erceg et al. (2000) or Kumhof and Laxton (2007b)); on the other hand, in the distribution stage, assembled goods (which are basically a composite tradable good) and non-tradable goods were assumed to feature a low substitutability as in Mendoza (2005) and Kumhof and Laxton (2007b). The distribution parameters of the production function in each stage were calibrated to match the NA import content and non-tradable goods' content of each type of final good. The degree of monopolistic competition among distributors was assumed to be lower than among manufacturers, with the markup being set to 5%, except in the case of exporters, where fiercer competition is likely to determine a lower markup. In terms of price stickiness, an average duration of price contracts of 2 quarters was assumed for all distributors except for exporters, whose prices are assumed to adjust slightly faster. Real rigidities related to the import content adjustment costs were set to ensure a smooth adjustment of import contents to real exchange rate fluctuations.

Government's average tax rates were calibrated to match the share of revenue-to-GDP ratio in the data. The same applies to EU transfers and to expenditure components (government consumption and investment and government transfers). The parameters of the fiscal policy rule were calibrated to impose a structural budget balance rule (unit fiscal policy stance parameter) and to ensure a smooth labour income tax rate adjustment. The target debt-to-GDP ratio was set to 53%, implying a fiscal balance-to-GDP ratio of -2.1% in the steady-state.<sup>19</sup>

As presented in Table 2, the steady-state calibration is, in general, able to match fairly reasonably the key ratios of the Portuguese economy and deliver a capital-to-GDP ratio that seems plausible by industry standards.<sup>20</sup> The most important aspect that arises is that the evolution of the Portuguese economy in the recent past is not fully consistent with a sustainable path, featuring a debt-to-GDP ratio below 60% and a reasonable net foreign debt-to-GDP ratio.

<sup>&</sup>lt;sup>19</sup>The values assumed for the debt-to-GDP target and the implied fiscal balance can be questioned in view of the medium term objective that has been set by the European Commission for Portugal (a structural budget balance of -0.5%, implying a debt-to-GDP ratio of close to 12%). However, since in the historical period that was used to calibrate the model the debt-to-GDP ratio averaged 57%, it does not seem reasonable to calibrate it to match something substantially different from historical figures.

<sup>&</sup>lt;sup>20</sup>Unfortunately, the Portuguese NA does not include figures for capital stock.

	Parameter	Value
Monetary union parameters		
Euro area interest rate (annualised)	$i^*$	1.05
Euro area labour-augmenting prod. growth (annualised)	g	1.02
Euro area inflation target (annualised)	$\pi^*$	1.02
Euro area EoS between domestic and imported goods	$\xi^*$	1.50
Households and Unions		
Households discount rate (annualised)	$\beta$	0.97
Intertemporal elasticity of substitution	$\frac{1}{\gamma}$	0.20
OLG households instant probability of death (annualised)	$1 - \dot{\theta}$	0.04
OLG households habit persistence	ν	0.70
Consumption share - OLG households	$\eta_{OLG}$	0.73
Consumption share - LIQ households	$\eta_{LIQ}$	0.70
Lifetime productivity decline rate (annualised)	$1-\chi$	0.04
Share of LIQ households	$\psi$	0.30
Share of dividend transfers from OLG to LIQ households	ι	0.15
Wage mark-up	$\frac{\sigma_U}{\sigma_U-1}$	1.25
Wage rigidity - Adjustment cost	$\phi_U$	200
Manufacturers		
Depreciation rate (annualised)	δ	0.09
EoS between capital and labour	$\xi_J$	0.99
Price markup - tradables	$\frac{\sigma_T}{\sigma_T - 1}$	1.10
Price markup - non-tradables	$\frac{\sigma_N}{\sigma_N}$	1.20
Capital adjustment cost	$\phi_{KJ}$	50
Investment adjustment cost	$\phi_{IJ}$	100
Price adjustment cost	$\phi_{PJ}$	200
Quasi labour income share - tradables	$\alpha_T$	0.21
Quasi labour income share - non-tradables	$lpha_N$	0.24
Distributors		
EoS domestic tradable/imported good	$\xi_{AF}$	1.50
EoS assembled/non-tradable good	$\xi_F$	0.50
Price markup (domestic distributors)	$\frac{\sigma_F}{\sigma_F-1}, F \neq X$	1.05
Price markup (exporters)	$\frac{\sigma_{F}}{\sigma_{W}}$	1.03
Import content adjustment cost	$\phi_{AF}$	2
Price adjustment cost	$\phi_{PF}$	200
Government		
Labour income tax rate	$ au_L$	0.29
Consumption tax rate	$ au_C$	0.30
Capital income tax rate	$ au_K$	0.17
Employers' social security contribution rate	$ au_{SP}$	0.19
Debt to GDP ratio (annualised)	$\frac{b}{adr}$	0.53
Fiscal stance parameter	$d_{damn}$	1.00
Speed adjustment towards the target debt ratio parameter	$d_{debt}$	0.10

## Table 1: Main parameters

	Data	Model
Expenditure (as a $\%$ of GDP)		
Private consumption	0.64	0.61
Government consumption and GFCF	0.22	0.21
Private investment	0.21	0.21
Exports	0.29	0.29
Imports	0.37	0.33
Labour income share (as a % of overall income)	0.57	0.56
Tradable goods	0.54	0.54
Non-tradable goods	0.58	0.58
Capital-output ratio (as a % of output)	NA	2.34
Tradable goods	NA	2.53
Non-tradable goods	NA	2.21
Government (as a % of GDP)		
Debt stock	0.57	0.53
Fiscal balance	-0.07	-0.02
Overall revenues	0.38	0.39
Overall expenditure	0.45	0.41
External account (as a $\%$ of GDP)		
Net foreign assets	-0.60	-0.60
Current account	-0.06	-0.02
Trade balance	-0.08	-0.04

Table 2: Steady-state key ratios

# 5 The macroeconomic impacts of increasing competition in the product and labour markets

In this section we illustrate the impact of reforms aimed at increasing competition in the nontradable goods and labour markets. The impact of specific reforms affecting markups is hardly measurable on a broad basis, implying a detailed assessment of the reforms at issue. International comparisons are not very helpful, since the impact of specific reforms depends crucially on the design and enforcement of the reform, on the existing legal framework, on the structure of the goods and labour markets and on the degree of openness of the economy. Since the assessment of the impact of specific reforms in wage and price markups is beyond the scope of this study, we assess the impact on the macroeconomic scenario of an increase in competition in the domestic markets by simulating shocks in non-tradable goods price and wage markups. We start by illustrating the impact of a decline in the non-tradable goods price markup, then we do the same for the wage markup and, finally we implement both shocks together. It should be highlighted, as already mentioned in Section 2, that there is evidence pointing to a positive correlation between nontradable goods price markups, though in our model these are assumed to be fully exogenous and, therefore, uncorrelated.

A number of aspects should be made clear from the outset. Firstly, all simulations are performed in a context of perfect foresight and perfect information, implying that the transition path is known with certainty by all agents from the outset.<sup>21</sup> These assumptions may affect the shortrun dynamics fastening the adjustment, but they are not likely to affect significantly the long-run impacts. A second important aspect which must be kept in mind is that the government follows a labour income tax based fiscal rule. The choice of the labour income tax rate as the endogenous policy tool is arbitrary and the long-run impact of the simulations depends on that assumption.

The macroeconomic impact of increasing competition in the non-tradable goods market is assessed through the simulation of a permanent decline of 10% in the markup over the marginal costs (corresponding to a decline of the markup from 20% to 18%). Since the non-tradable goods markup is fully determined by the elasticity of substitution between varieties of the non-tradable good,  $\sigma_{N,t}$ , and given that the shock follows an autoregressive process, the impact gradually feeds through the economy. The persistence parameter implies that roughly 80% of the decline is achieved in two years. The results are reported in the first column of Table 3, as percentage deviations from the baseline level unless otherwise indicated.

*Ceteris paribus* the permanent decline in the non-tradable goods markup translates into: (i) a decline in final goods prices, including exports, and a real exchange rate depreciation; (ii) a decline

<sup>&</sup>lt;sup>21</sup>The model is simulated in Portable TROLL and is solved using the Newton-based perfect foresight simulation algorithm proposed in Juillard, Laxton, McAdam and Pioro (1998).

in the capital component of human wealth, reflecting lower dividends from non-tradable goods' manufacturers; (iii) a decline in the relative price of non-tradable goods with respect to tradable, implying a change in factor intensities of final goods. However, in a dynamic general equilibrium model the impact of the shock goes well beyond mechanical impacts, reflecting spillover effects and the adjustment dynamics implied by nominal and real rigidities. Moreover, the income tax based fiscal policy rule may significantly affect the dynamics of households' behaviour through the consumption/leisure choice.

The decline in the non-tradable goods markup has important spillovers to sectors that use this good as an intermediate input. The marginal costs of production of final goods decline across the board, implying among other effects a real exchange rate depreciation and an increase in international competitiveness of Portuguese production. The real exchange rate depreciation drives export demand upwards and fosters the substitution of imports by domestic tradable goods, implying a higher demand for intermediate goods.

The increase in demand for intermediate goods boosts demand for capital and labour services. To provide a higher labour supply, households demand higher labour income, implying an increase in households' real wage or a decline in the labour income tax. The simulation results show that real households' wages are driven upwards, since the decline in the labour income tax does not shift labour supply enough. Investment goods become cheaper as costs related to the non-tradable goods component decline. Therefore, in the new steady-state hours worked are higher and the same applies to the capital stock; however, firms substitute labour for capital services as much as their technological constraint allows them to, since capital is cheaper in relative terms, implying that intermediate goods production technology becomes more capital intensive.

In the new steady-state, consumption is significantly higher mainly as a result of the impact on households' wealth of a higher wage rate and a lower labour income tax rate. In addition, financial wealth also increases due to enhanced competitiveness conditions, which improve the trade balance and the net foreign debt position. All in all, the negative mechanical impact in wealth, stemming from the decline in dividends, is more than offset by higher after-tax wage income and improved net foreign position implying a positive long-run impact on households' consumption.

An apparently puzzling fact is that the decline in the non-tradable goods price markup boosts demand for tradable goods more than for non-tradable goods, when the opposite might seem more intuitive. There are mainly three reasons for this: (i) the increase in non-tradable goods intensity of final goods, in a context where non-tradable goods get relatively cheaper, is limited by the low elasticity of substitution of assembled good for non-tradables; (ii) in a small open economy integrated in a monetary union, the real exchange rate tends to be crucial in terms of adjustment, with a direct effect on exports, which are tradable goods intensive, and import contents, implying a strong substitution of imports for domestic tradable goods; and (iii) the most intensive final good in non-tradable goods is government consumption, where the assumption is that demand remains unchanged.

Similarly to non-tradable goods, the increase in labour market competition is assessed through the simulation of a permanent decline of 10% in the steady-state wage markup, which amounts to a cut in the wage premia over the marginal rate of substitution of consumption for labour by 2.5 percentage points (from 25% to 22.5%). As in the previous case, the persistence parameter of the shock implies that roughly 80% of the markup decline is achieved in two years. The results are reported in the second column of Table 3.

The permanent decline in the wage markup translates *ceteris paribus* into: (i) a decline in the relative price of labour services with respect to capital from the manufacturers' point of view; (ii) a decline in labour income, due to lower labour union dividends, from the households' perspective; and (iii) lower marginal costs, favouring a decline in intermediate and final goods prices and a real exchange rate depreciation.

The increase in competition in the labour market lowers the relative price of labour with respect to capital, inducing firms to adopt more labour intensive technologies, and driving labour demand upwards. As previously mentioned, the increased labour demand can only be matched by a higher labour supply at the cost of a higher households real wage income.<sup>22</sup> Since the impact of lower labour income tax does not shift labour supply enough, households' real wage will increase.

Turning to the evolution of prices and competitiveness conditions of the economy, the decline in the wage markup implies lower firm wage rate and lower marginal cost for manufacturers. In a context of exogenous steady-state price markups, all declines in marginal costs are fully passed on to costumers, implying that distributors will benefit from the decline in wage markups. Since the labour content of tradable and non-tradable goods is not very different, the impact on prices is similar and, therefore, the differentiated impact in final goods mainly reflects different import contents. As in the non-tradables case, the real exchange rate depreciation implied by lower domestic costs has a significant impact on competitiveness conditions, implying not only permanent market share gains in the export market, but also a substitution of imports for domestic tradable goods, implying an improvement in the current account and net foreign debt position of the economy.

The increase in labour market competition is also likely to be beneficial for households, since it yields higher consumption levels. Households' consumption increases substantially in the new steady-state, reflecting the positive impact of the shock both in the households' human and financial wealth. The negative impact of the shock in households' labour income that might be suggested *ceteris paribus* is fully misleading, since general equilibrium effects imply that the decline in labour

 $<sup>^{22}</sup>$ It should be mentioned that the decrease in wage markup *per se* does not affect labour supply since the after-tax households' real wage income does not include labour union dividends.

unions' dividends is more than compensated by higher households' real wage and lower labour income taxes as mentioned above, determining a higher human wealth. In addition, as in the non-tradables case, the improvement in the economy's net foreign debt position affects financial wealth positively.

The impact of a package to increase competition in the domestic markets, i.e. a simultaneous increase in non-tradable goods and labour market competition, raises a number of issues on the timing and implementation of this type of reforms. The results are reported in the third column of Table 3.

The first issue is that the impact of the shock is additive, meaning that the joint impact of decreasing the markup in non-tradable goods prices and wages can be reasonably captured as the sum of the impact of the individual reforms. The policy implication seems to be that from a purely economic point of view there are no big gains or losses from implementing both reform packages simultaneously. However, it should be kept in mind that we have assumed that product and labour market reforms are fully independent, and this, according to Jean and Nicoletti (2002), may not be very realistic, since the wage premia may reflect a share of monopoly rents generated by lack of competition in the goods market and, therefore, higher competition in goods market may induce by itself a decline in the wage markups.

A second issue is related with the transition path of these reforms. The short-run impact of higher competition in product and labour markets translates into a negative effect in households' consumption (see Figure 1). This impact reflects, on the one hand, the real interest rate profile, which is a particular feature of the adjustment of a small open economy integrated in a monetary union, and, on the other hand, the impact of lower dividend income arising from the decline in nontradable goods markup. With respect to the profile of real interest rate, it should be noted that the reduction in markups implies that the economy goes through a temporary period of inflation below the baseline level. Since a small open economy integrated in a monetary union does not affect monetary policy decisions, it implies a temporary increase in the real interest rate, which translates into a decline in human wealth and in the marginal propensity to consume out of wealth (in the transition path real interest rates induce OLG households to postpone consumption expenditures, since they get a higher real reward for saving in the current period). However, it should be clear that, on the one hand, the negative impact on consumption is a short-run effect, since real interest rates are not permanently affected by this shock, and, on the other hand, as consumers are forward looking they know that in the future these reforms will have a positive impact on their consumption. Therefore, this temporary negative effect should not be a severe political obstacle hindering the implementation of these reforms. Nevertheless, the policy implication one may derive is that the implementation of these reforms in the peak of the business cycle when inflation, consumption and economic activity stand well above the steady-state level, will not only yield the long-run beneficial

	NT	W	NT + W
Expenditure			
CDP	1 2 2	1 16	2.45
ODI Drivata consumption	1.00	1.10	2.40
Invostment	2.08	0.82	2.88
Exports	1.00	1.00	2.00
Imports	0.31	0.30	0.61
Wages and real exchange rate			
Real wage rate (firms)	0.54	-0.59	-0.02
Labour income tax (pp.)	-2.01	-1.24	-3.16
Real exchange rate $(depreciation(+)/appreciation(-))$	1.53	0.75	2.25
Wealth and bond stocks			
Total human wealth	1.23	1.17	2.33
Government bonds (as a $\%$ of GDP)	0.02	0.01	0.03
For eign bonds (as a $\%$ of GDP)	4.70	4.48	9.19
Supply and labour market conditions			
Tradable goods sector			
Output	1.99	1.73	3.66
Hours worked	1.72	2.04	3.67
Price	0.26	-0.28	-0.01
Non-tradable goods sector			
Output	1.44	0.81	2.23
Hours worked	1.20	1.08	2.24
Price	-1.38	-0.31	-1.67

Table 3: Impact of an increase in competition in the domestic market

impacts already mentioned but will also contribute to dampen business cycle fluctuation.

Finally, and more importantly, the simulation results suggests that increasing competition in domestic markets clearly fosters economic growth prospects, by improving exports' competitiveness, increasing employment and promoting a more efficient allocation of resources. Furthermore, the macroeconomic impact presented in this study tends to be on the low side, since reforms aimed at increasing competition are also likely to induce important productivity gains, stemming from the fact that fiercer competition tends to lead to the shutdown of the less efficient production units, to a more efficient organisation of the remaining ones and to the installation of new and more productive units, for example through foreign direct investment inflows, and to higher R&D efforts, in line with Blanchard and Giavazzi (2003), Aghion et al. (2007) and OECD (2008).

Figure 1: Impact of an increase in competition in the domestic market (percentage changes from baseline levels except real interest rate)







 Human wealth -Hours

--

## 6 Conclusions and directions for further research

In this study we assess the macroeconomic impact of reforms aimed at increasing product and labour market competition in Portugal, using a multi-sectoral dynamic general equilibrium model. We find that these reforms can be valuable instruments to promote necessary adjustments in the Portuguese economy, within the monetary union framework, mainly through important international competitiveness gains generated by a real exchange rate adjustment, though in a less dramatic way than the sharp nominal wage rate cut suggested in Blanchard (2007).

Specifically, our results suggest that a 10% decrease in non-tradable goods prices and wage markups has non-negligible positive impacts on economic activity, households' consumption and hours worked, reflecting improved international competitiveness conditions. One should highlight that these impacts are likely to be underestimated, since a number of relevant aspects are not featured in the model. In particular, as pointed out in Blanchard and Giavazzi (2003), Aghion et al. (2007) and in OECD (2008), the suggested reforms tend to induce foreign direct investment inflows and higher R&D effort, which typically lead to an increase in total factor productivity that is not being accounted for.

Our results also suggest that the deflationary short-run impact of the reforms under consideration temporarily drives the real interest rate upwards, implying a negative impact on households' consumption and on economic activity. This impact results from the fact that euro area monetary policy decisions are not affected by small country specific shocks. However, it should be clear that this is a temporary effect, which should not be a severe political obstacle for the implementation of these type of reforms. Nevertheless, the policy implication one may recover is that the implementation of these reforms at the peak of the business cycle will not only yield long-run beneficial impacts but will also contribute to dampen business cycle fluctuation.

The results obtained leave a number of directions for further research not only with respect to the issue of domestic competition, but also on the development of *PESSOA*. In terms of the issue of domestic competition, it would be important to develop the implementation of shocks in a way that allows for an appropriate capturing of the correlation between reforms in product and labour market and productivity gains. As far as *PESSOA* is concerned, the brand new general equilibrium model calibrated for the Portuguese economy, we plan to use it, in the near future, to assess the macroeconomic impact of shocks that have hit the Portuguese economy since the beginning of the euro and draw some policy implications, similarly to what has been done here for the issue of competition in the domestic markets. A more ambitious project would be to estimate a streamlined version of the model using Bayesian techniques and use it to perform forecasts for the Portuguese economy.

## References

- Adolfson, M., Laseén, S., Lindé, J. and Villani, M. (2005), Bayesian estimation of an open economy DSGE model with incomplete exchange rate pass-through, Working Paper n.<sup>o</sup> 179, Sveriges Riksbank.
- Aghion, P., Askenazy, P., Bouelès, R., Cette, G. and Dromel, N. (2007), Education, market rigidities and growth, Discussion Paper n.<sup>o</sup> 3166, IZA.
- Almeida, V. and Félix, R. (2006), Computing potential output and the output-gap for the Portuguese economy, Economic Bulletin, Banco de Portugal.
- Bayoumi, T., Laxton, D. and Pesenti, P. (2004), Benefits and spillovers of greater competition in the Europe: a macroeconomic assessment, Working Paper n.º 341, European Central Bank, Frankfurt, Germany.
- Blanchard, O. (1985), 'Debts, deficits and finite horizons', Journal of Political Economy 93(2), 223– 247.
- Blanchard, O. (2007), 'Adjusting within the euro. the difficult case of portugal.', Portuguese Economic Journal 6, 1–21.
- Blanchard, O. and Fischer, S. (1989), Lectures on Macroeconomics, MIT Press, Cambridge, Massachussets.
- Blanchard, O. and Giavazzi, F. (2003), 'Macroeconomic effects of regulation and deregulation in goods and labour markets', *Quarterly Journal of Economics* 118(3), 879–907.
- Castro, G. (2006), Consumption, disposable income and liquidity constraints, Economic Bulletin, Banco de Portugal.
- Coenen, G., McAdam, P. and Straub, R. (2007), Tax reform and labour-market performance in the euro area: a simulation-based analysis using the new area-wide model, Working Paper n.º 747, European Central Bank, Frankfurt, Germany.
- Conway, P., Janod, V. and Nicoletti, G. (2005), Product market regulation in OECD countries: 1998 to 2003, Economics Department Working Paper n.<sup>o</sup> 419, OECD.
- Conway, P. and Nicoletti, G. (2006), Product market regulation in the non-manufacturing sectors of OECD countries, Economics Department Working Paper n.<sup>o</sup> 530, OECD.
- Erceg, C. J., Henderson, D. W. and Levin, A. T. (2000), 'Optimal monetary policy with staggered wage and price contracts', *Journal of Monetary Economics* 46(2), 281–313.

- European Commission (2005), The economic costs of non-Lisbon, Occasional Paper n.º 16, European Commission.
- European Commission (2008), EMU@10:successes and challenges after 10 years of Economic and Monetary Union, Economic Policy n.º 2, European Commission.
- Fagan, G., Gaspar, V. and Pereira, A. (2004), 'The New Keynesian Microfoundations', pp. 69–104.
- Forni, L., Gerali, A. and Pisano, M. (2008), Macroeonomic effects of greater competition in the service sector: the case of italy, Working Paper, Banca di Italia.
- Frenkel, J. and Razin, A. (1996), Fiscal Policies and Growth in the World Economy, 3 edn, MIT Press, Cambridge, Massachussets.
- Harrison, R., Nikolov, K., Quinn, M., Ramsay, G., Scott, A. and Thomas, R. (2005), The Bank of England Quarterly Model, Bank of England.
- Høj, J. (2007), Competition law and policy indicators, Economics Department Working Paper n.° 568, OECD.
- Høj, J., Jimenez, M., Maher, M., Nicoletti, G. and Wise, M. (2007), Product market competition in OECD countries: taking stock and moving forward, Economics Department Working Paper n.º 575, OECD.
- Jean, S. and Nicoletti, G. (2002), Product market regulation and wage premia in Europe and North America, Economics Department Working Paper n.<sup>o</sup> 318, OECD.
- Juillard, M., Laxton, D., McAdam, P. and Pioro, H. (1998), 'An algorithm competition: Firstorder iteration versus Newton-based techniques', *Journal of Economic Dynamics and Control* 22, 1291–1318.
- Kilponen, J., Kinnunen, H. and Ripatti, A. (2006a), Demographic uncertainty and labour market imperfections in small open economy, Computing in Economics and Finance 227, Society for Computational Economics.
- Kilponen, J., Kinnunen, H. and Ripatti, A. (2006b), Population ageing in a small open economy
   some policy experiments with a tractable general equilibrium model, Discussion Paper 28/2006, Bank of Finland, Helsinki.
- Kilponen, J. and Ripatti, A. (2006), Labour and product market competition in a small open economy: Simulation results using a DGE model of the Finnish economy, Discussion Paper 5/2006, Bank of Finland, Helsinki.

- Kim, J. (2000), 'Constructing and estimating as realistic optimizing model of monetary policy', Journal of Monetary Economics 45, 329–359.
- Kumhof, M. and Laxton, D. (2007*a*), The Global Integrated Monetary and Fiscal Model, *in mimeo*, International Monetary Fund.
- Kumhof, M. and Laxton, D. (2007b), A party without a hangover ?, Research Paper n.º 43, International Monetary Fund.
- Kumhof, M. and Laxton, D. (2008), Chile's Structural Fiscal Surplus Rule: a model based evaluation, Research Paper forthcoming, International Monetary Fund.
- Laxton, D. and Pesenti, P. (2003), 'Monetary rules for small, open, emerging economies', Journal of Monetary Economics 50, 1109–1146.
- Mendoza, E. (2005), Real exchange rate volatility and the price of non-tradables in sudden-stopprone economies, Working Paper 11691, NBER, Cambridge, Massachussets: National Bureau of Economic Research.
- Musso, A. and Westermann, T. (2005), Assessing potential output in the euro area: a growth accounting perspective, Occasional paper n.º 22, European Central Bank, Frankfurt, Germany.
- Nicoletti, G. and Scarpetta, S. (2005), Product market reforms and employment in OECD countries, Economics Department Working Paper 472, OECD.
- OECD (2006), Economic Survey: Portugal, OECD, chapter 5, pp. 114–132.
- OECD (2008), Economic Survey: Portugal, OECD, chapter 3, pp. 69–115.
- Proietti, T. and Musso, A. (2007), Growth accounting for the euro area: a structural approach, Working Paper n.<sup>o</sup> 804, European Central Bank, Frankfurt, Germany.
- Rotemberg, J. J. (1987), *The New Keynesian Microfoundations*, Stanley Fischer ed., MIT Press, pp. 69–104.
- Woodford, M. (2003), Interest and Prices: Foundations of a Theory of Monetary Policy, Princeton University Press.
- Yaari, M. (1965), 'Uncertain lifetime, life insurance and the theory of the consumer', Review of Economic Studies 32(2), 137–150.

## **WORKING PAPERS**

## 2000

1/00	UNEMPLOYMENT DURATION: COMPETING AND DEFECTIVE RISKS — John T. Addison, Pedro Portugal
2/00	THE ESTIMATION OF RISK PREMIUM IMPLICIT IN OIL PRICES — Jorge Barros Luís
3/00	EVALUATING CORE INFLATION INDICATORS — Carlos Robalo Marques, Pedro Duarte Neves, Luís Morais Sarmento
4/00	LABOR MARKETS AND KALEIDOSCOPIC COMPARATIVE ADVANTAGE — Daniel A. Traça
5/00	WHY SHOULD CENTRAL BANKS AVOID THE USE OF THE UNDERLYING INFLATION INDICATOR? — Carlos Robalo Marques, Pedro Duarte Neves, Afonso Gonçalves da Silva
6/00	USING THE ASYMMETRIC TRIMMED MEAN AS A CORE INFLATION INDICATOR — Carlos Robalo Marques, João Machado Mota
	2001
1/01	THE SURVIVAL OF NEW DOMESTIC AND FOREIGN OWNED FIRMS — José Mata, Pedro Portugal
2/01	GAPS AND TRIANGLES — Bernardino Adão, Isabel Correia, Pedro Teles
3/01	A NEW REPRESENTATION FOR THE FOREIGN CURRENCY RISK PREMIUM — Bernardino Adão, Fátima Silva
4/01	ENTRY MISTAKES WITH STRATEGIC PRICING — Bernardino Adão
5/01	FINANCING IN THE EUROSYSTEM: FIXED VERSUS VARIABLE RATE TENDERS — Margarida Catalão-Lopes
6/01	AGGREGATION, PERSISTENCE AND VOLATILITY IN A MACROMODEL — Karim Abadir, Gabriel Talmain
7/01	SOME FACTS ABOUT THE CYCLICAL CONVERGENCE IN THE EURO ZONE — Frederico Belo
8/01	TENURE, BUSINESS CYCLE AND THE WAGE-SETTING PROCESS — Leandro Arozamena, Mário Centeno
9/01	USING THE FIRST PRINCIPAL COMPONENT AS A CORE INFLATION INDICATOR — José Ferreira Machado, Carlos Robalo Marques, Pedro Duarte Neves, Afonso Gonçalves da Silva
10/01	IDENTIFICATION WITH AVERAGED DATA AND IMPLICATIONS FOR HEDONIC REGRESSION STUDIES — José A.F. Machado, João M.C. Santos Silva

	2002
1/02	QUANTILE REGRESSION ANALYSIS OF TRANSITION DATA — José A.F. Machado, Pedro Portugal
2/02	SHOULD WE DISTINGUISH BETWEEN STATIC AND DYNAMIC LONG RUN EQUILIBRIUM IN ERROR CORRECTION MODELS? — Susana Botas, Carlos Robalo Marques
3/02	MODELLING TAYLOR RULE UNCERTAINTY — Fernando Martins, José A. F. Machado, Paulo Soares Esteves
4/02	PATTERNS OF ENTRY, POST-ENTRY GROWTH AND SURVIVAL: A COMPARISON BETWEEN DOMESTIC AND FOREIGN OWNED FIRMS — José Mata, Pedro Portugal
5/02	BUSINESS CYCLES: CYCLICAL COMOVEMENT WITHIN THE EUROPEAN UNION IN THE PERIOD 1960-1999. A FREQUENCY DOMAIN APPROACH — João Valle e Azevedo
6/02	AN "ART", NOT A "SCIENCE"? CENTRAL BANK MANAGEMENT IN PORTUGAL UNDER THE GOLD STANDARD, 1854 -1891 — Jaime Reis
7/02	MERGE OR CONCENTRATE? SOME INSIGHTS FOR ANTITRUST POLICY — Margarida Catalão-Lopes
8/02	DISENTANGLING THE MINIMUM WAGE PUZZLE: ANALYSIS OF WORKER ACCESSIONS AND SEPARATIONS FROM A LONGITUDINAL MATCHED EMPLOYER-EMPLOYEE DATA SET — Pedro Portugal, Ana Rute Cardoso
9/02	THE MATCH QUALITY GAINS FROM UNEMPLOYMENT INSURANCE — Mário Centeno
10/02	HEDONIC PRICES INDEXES FOR NEW PASSENGER CARS IN PORTUGAL (1997-2001) — Hugo J. Reis, J.M.C. Santos Silva
11/02	THE ANALYSIS OF SEASONAL RETURN ANOMALIES IN THE PORTUGUESE STOCK MARKET — Miguel Balbina, Nuno C. Martins
12/02	DOES MONEY GRANGER CAUSE INFLATION IN THE EURO AREA? — Carlos Robalo Marques, Joaquim Pina
13/02	INSTITUTIONS AND ECONOMIC DEVELOPMENT: HOW STRONG IS THE RELATION? — Tiago V.de V. Cavalcanti, Álvaro A. Novo
	2003
1/03	FOUNDING CONDITIONS AND THE SURVIVAL OF NEW FIRMS — P.A. Geroski, José Mata, Pedro Portugal
2/03	THE TIMING AND PROBABILITY OF FDI: AN APPLICATION TO THE UNITED STATES MULTINATIONAL ENTERPRISES — José Brandão de Brito, Felipa de Mello Sampayo
3/03	OPTIMAL FISCAL AND MONETARY POLICY: EQUIVALENCE RESULTS — Isabel Correia, Juan Pablo Nicolini, Pedro Teles

4/03	FORECASTING EURO AREA AGGREGATES WITH BAYESIAN VAR AND VECM MODELS — Ricardo Mourinho Félix, Luís C. Nunes
5/03	CONTAGIOUS CURRENCY CRISES: A SPATIAL PROBIT APPROACH — Álvaro Novo
6/03	THE DISTRIBUTION OF LIQUIDITY IN A MONETARY UNION WITH DIFFERENT PORTFOLIO RIGIDITIES — <i>Nuno Alves</i>
7/03	COINCIDENT AND LEADING INDICATORS FOR THE EURO AREA: A FREQUENCY BAND APPROACH — António Rua, Luís C. Nunes
8/03	WHY DO FIRMS USE FIXED-TERM CONTRACTS? — José Varejão, Pedro Portugal
9/03	NONLINEARITIES OVER THE BUSINESS CYCLE: AN APPLICATION OF THE SMOOTH TRANSITION AUTOREGRESSIVE MODEL TO CHARACTERIZE GDP DYNAMICS FOR THE EURO-AREA AND PORTUGAL — Francisco Craveiro Dias
10/03	WAGES AND THE RISK OF DISPLACEMENT — Anabela Carneiro, Pedro Portugal
11/03	SIX WAYS TO LEAVE UNEMPLOYMENT — Pedro Portugal, John T. Addison
12/03	EMPLOYMENT DYNAMICS AND THE STRUCTURE OF LABOR ADJUSTMENT COSTS — José Varejão, Pedro Portugal
13/03	THE MONETARY TRANSMISSION MECHANISM: IS IT RELEVANT FOR POLICY? — Bernardino Adão, Isabel Correia, Pedro Teles
14/03	THE IMPACT OF INTEREST-RATE SUBSIDIES ON LONG-TERM HOUSEHOLD DEBT: EVIDENCE FROM A LARGE PROGRAM — Nuno C. Martins, Ernesto Villanueva
15/03	THE CAREERS OF TOP MANAGERS AND FIRM OPENNESS: INTERNAL VERSUS EXTERNAL LABOUR MARKETS — Francisco Lima, Mário Centeno
16/03	TRACKING GROWTH AND THE BUSINESS CYCLE: A STOCHASTIC COMMON CYCLE MODEL FOR THE EURO AREA — João Valle e Azevedo, Siem Jan Koopman, António Rua
17/03	CORRUPTION, CREDIT MARKET IMPERFECTIONS, AND ECONOMIC DEVELOPMENT — António R. Antunes, Tiago V. Cavalcanti
18/03	BARGAINED WAGES, WAGE DRIFT AND THE DESIGN OF THE WAGE SETTING SYSTEM — Ana Rute Cardoso, Pedro Portugal
19/03	UNCERTAINTY AND RISK ANALYSIS OF MACROECONOMIC FORECASTS: FAN CHARTS REVISITED — Álvaro Novo, Maximiano Pinheiro
	2004
1/04	HOW DOES THE UNEMPLOYMENT INSURANCE SYSTEM SHAPE THE TIME PROFILE OF JOBLESS DURATION? — John T. Addison, Pedro Portugal

2/04	REAL EXCHANGE RATE AND HUMAN CAPITAL IN THE EMPIRICS OF ECONOMIC GROWTH — Delfim Gomes Neto
3/04	ON THE USE OF THE FIRST PRINCIPAL COMPONENT AS A CORE INFLATION INDICATOR — José Ramos Maria
4/04	OIL PRICES ASSUMPTIONS IN MACROECONOMIC FORECASTS: SHOULD WE FOLLOW FUTURES MARKET EXPECTATIONS? — Carlos Coimbra, Paulo Soares Esteves
5/04	STYLISED FEATURES OF PRICE SETTING BEHAVIOUR IN PORTUGAL: 1992-2001 — Mónica Dias, Daniel Dias, Pedro D. Neves
6/04	A FLEXIBLE VIEW ON PRICES — Nuno Alves
7/04	ON THE FISHER-KONIECZNY INDEX OF PRICE CHANGES SYNCHRONIZATION — D.A. Dias, C. Robalo Marques, P.D. Neves, J.M.C. Santos Silva
8/04	INFLATION PERSISTENCE: FACTS OR ARTEFACTS? — Carlos Robalo Marques
9/04	WORKERS' FLOWS AND REAL WAGE CYCLICALITY — Anabela Carneiro, Pedro Portugal
10/04	MATCHING WORKERS TO JOBS IN THE FAST LANE: THE OPERATION OF FIXED-TERM CONTRACTS — José Varejão, Pedro Portugal
11/04	THE LOCATIONAL DETERMINANTS OF THE U.S. MULTINATIONALS ACTIVITIES — José Brandão de Brito, Felipa Mello Sampayo
12/04	KEY ELASTICITIES IN JOB SEARCH THEORY: INTERNATIONAL EVIDENCE — John T. Addison, Mário Centeno, Pedro Portugal
13/04	RESERVATION WAGES, SEARCH DURATION AND ACCEPTED WAGES IN EUROPE — John T. Addison, Mário Centeno, Pedro Portugal
14/04	THE MONETARY TRANSMISSION N THE US AND THE EURO AREA: COMMON FEATURES AND COMMON FRICTIONS — Nuno Alves
15/04	NOMINAL WAGE INERTIA IN GENERAL EQUILIBRIUM MODELS — Nuno Alves
16/04	MONETARY POLICY IN A CURRENCY UNION WITH NATIONAL PRICE ASYMMETRIES — Sandra Gomes
17/04	NEOCLASSICAL INVESTMENT WITH MORAL HAZARD — João Ejarque
18/04	MONETARY POLICY WITH STATE CONTINGENT INTEREST RATES — Bernardino Adão, Isabel Correia, Pedro Teles
19/04	MONETARY POLICY WITH SINGLE INSTRUMENT FEEDBACK RULES — Bernardino Adão, Isabel Correia, Pedro Teles
20/04	ACOUNTING FOR THE HIDDEN ECONOMY: BARRIERS TO LAGALITY AND LEGAL FAILURES — António R. Antunes, Tiago V. Cavalcanti

### 2005 1/05 SEAM: A SMALL-SCALE EURO AREA MODEL WITH FORWARD-LOOKING ELEMENTS - José Brandão de Brito, Rita Duarte 2/05 FORECASTING INFLATION THROUGH A BOTTOM-UP APPROACH: THE PORTUGUESE CASE — Cláudia Duarte, António Rua 3/05 USING MEAN REVERSION AS A MEASURE OF PERSISTENCE - Daniel Dias, Carlos Robalo Marques 4/05 HOUSEHOLD WEALTH IN PORTUGAL: 1980-2004 — Fátima Cardoso, Vanda Geraldes da Cunha 5/05 ANALYSIS OF DELINQUENT FIRMS USING MULTI-STATE TRANSITIONS — António Antunes 6/05 PRICE SETTING IN THE AREA: SOME STYLIZED FACTS FROM INDIVIDUAL CONSUMER PRICE DATA — Emmanuel Dhyne, Luis J. Álvarez, Hervé Le Bihan, Giovanni Veronese, Daniel Dias, Johannes Hoffmann, Nicole Jonker, Patrick Lünnemann, Fabio Rumler, Jouko Vilmunen INTERMEDIATION COSTS, INVESTOR PROTECTION AND ECONOMIC DEVELOPMENT 7/05 — António Antunes, Tiago Cavalcanti, Anne Villamil 8/05 TIME OR STATE DEPENDENT PRICE SETTING RULES? EVIDENCE FROM PORTUGUESE MICRO DATA - Daniel Dias, Carlos Robalo Margues, João Santos Silva 9/05 BUSINESS CYCLE AT A SECTORAL LEVEL: THE PORTUGUESE CASE - Hugo Reis 10/05 THE PRICING BEHAVIOUR OF FIRMS IN THE EURO AREA: NEW SURVEY EVIDENCE - S. Fabiani, M. Druant, I. Hernando, C. Kwapil, B. Landau, C. Loupias, F. Martins, T. Mathä, R. Sabbatini, H. Stahl, A. Stokman 11/05 CONSUMPTION TAXES AND REDISTRIBUTION - Isabel Correia UNIQUE EQUILIBRIUM WITH SINGLE MONETARY INSTRUMENT RULES 12/05 — Bernardino Adão, Isabel Correia, Pedro Teles A MACROECONOMIC STRUCTURAL MODEL FOR THE PORTUGUESE ECONOMY 13/05 — Ricardo Mourinho Félix THE EFFECTS OF A GOVERNMENT EXPENDITURES SHOCK 14/05 - Bernardino Adão, José Brandão de Brito MARKET INTEGRATION IN THE GOLDEN PERIPHERY - THE LISBON/LONDON EXCHANGE, 1854-1891 15/05 — Rui Pedro Esteves, Jaime Reis, Fabiano Ferramosca 2006 1/06 THE EFFECTS OF A TECHNOLOGY SHOCK IN THE EURO AREA - Nuno Alves, José Brandão de Brito, Sandra Gomes, João Sousa 2/02 THE TRANSMISSION OF MONETARY AND TECHNOLOGY SHOCKS IN THE EURO AREA - Nuno Alves, José Brandão de Brito, Sandra Gomes, João Sousa

4/06 THE PRICE SETTING BEHAVIOU — Fernando Martins	R OF PORTUGUESE FIRMS EVIDENCE FROM SURVEY DATA
5/06 STICKY PRICES IN THE EURO A — L. J. Álvarez, E. Dhyne, M. H H. Stahl, P. Vermeulen and J	REA: A SUMMARY OF NEW MICRO EVIDENCE oeberichts, C. Kwapil, H. Le Bihan, P. Lünnemann, F. Martins, R. Sabbatini, I. Vilmunen
6/06 NOMINAL DEBT AS A BURDEN C — Javier Díaz-Giménez, Giorgia	DN MONETARY POLICY a Giovannetti , Ramon Marimon, Pedro Teles
7/06 A DISAGGREGATED FRAMEWO FINANCES — Jana Kremer, Cláudia Rodrig	RK FOR THE ANALYSIS OF STRUCTURAL DEVELOPMENTS IN PUBLIC nues Braz, Teunis Brosens, Geert Langenus, Sandro Momigliano, Mikko
Spolander	
8/06 IDENTIFYING ASSET PRICE BOO — José A. F. Machado, João So	DMS AND BUSTS WITH QUANTILE REGRESSIONS Dusa
9/06 EXCESS BURDEN AND THE COS — António Afonso, Vítor Gaspa	ST OF INEFFICIENCY IN PUBLIC SERVICES PROVISION
10/06 MARKET POWER, DISMISSAL TH FORCES IN WAGE BARGAINING — Anabela Carneiro, Pedro Pol	HREAT AND RENT SHARING: THE ROLE OF INSIDER AND OUTSIDER
11/06 MEASURING EXPORT COMPETI THE EURO AREA COUNTRIES — Paulo Soares Esteves, Caroi	TIVENESS: REVISITING THE EFFECTIVE EXCHANGE RATE WEIGHTS FOR
12/06 THE IMPACT OF UNEMPLOYME ON MATCH QUALITY DISTRIBUT — Mário Centeno, Alvaro A. No	NT INSURANCE GENEROSITY TION VO
13/06 U.S. UNEMPLOYMENT DURATIO — José A.F. Machado, Pedro P	N: HAS LONG BECOME LONGER OR SHORT BECOME SHORTER? ortugal e Juliana Guimarães
14/06 EARNINGS LOSSES OF DISPLAG DATA SET — Anabela Carneiro, Pedro Por	CED WORKERS: EVIDENCE FROM A MATCHED EMPLOYER-EMPLOYEE
15/06 COMPUTING GENERAL EQUILIE — António Antunes, Tiago Cava	RIUM MODELS WITH OCCUPATIONAL CHOICE AND FINANCIAL FRICTIONS
16/06 ON THE RELEVANCE OF EXCHA — Bernardino Adao, Isabel Con	NGE RATE REGIMES FOR STABILIZATION POLICY reia, Pedro Teles
17/06 AN INPUT-OUTPUT ANALYSIS: L — Hugo Reis, António Rua	INKAGES VS LEAKAGES
2007	
1/07 RELATIVE EXPORT STRUCTURI — João Amador, Sónia Cabral,	ES AND VERTICAL SPECIALIZATION: A SIMPLE CROSS-COUNTRY INDEX José Ramos Maria

2/07	THE FORWARD PREMIUM OF EURO INTEREST RATES — Sónia Costa, Ana Beatriz Galvão
3/07	ADJUSTING TO THE EURO — Gabriel Fagan, Vítor Gaspar
4/07	SPATIAL AND TEMPORAL AGGREGATION IN THE ESTIMATION OF LABOR DEMAND FUNCTIONS — José Varejão, Pedro Portugal
5/07	PRICE SETTING IN THE EURO AREA: SOME STYLISED FACTS FROM INDIVIDUAL PRODUCER PRICE DATA — Philip Vermeulen, Daniel Dias, Maarten Dossche, Erwan Gautier, Ignacio Hernando, Roberto Sabbatini, Harald Stahl
6/07	A STOCHASTIC FRONTIER ANALYSIS OF SECONDARY EDUCATION OUTPUT IN PORTUGAL — Manuel Coutinho Pereira, Sara Moreira
7/07	CREDIT RISK DRIVERS: EVALUATING THE CONTRIBUTION OF FIRM LEVEL INFORMATION AND OF MACROECONOMIC DYNAMICS — Diana Bonfim
8/07	CHARACTERISTICS OF THE PORTUGUESE ECONOMIC GROWTH: WHAT HAS BEEN MISSING? — João Amador, Carlos Coimbra
9/07	TOTAL FACTOR PRODUCTIVITY GROWTH IN THE G7 COUNTRIES: DIFFERENT OR ALIKE? — João Amador, Carlos Coimbra
10/07	IDENTIFYING UNEMPLOYMENT INSURANCE INCOME EFFECTS WITH A QUASI-NATURAL EXPERIMENT — Mário Centeno, Alvaro A. Novo
11/07	HOW DO DIFFERENT ENTITLEMENTS TO UNEMPLOYMENT BENEFITS AFFECT THE TRANSITIONS FROM UNEMPLOYMENT INTO EMPLOYMENT — John T. Addison, Pedro Portugal
12/07	INTERPRETATION OF THE EFFECTS OF FILTERING INTEGRATED TIME SERIES — João Valle e Azevedo
13/07	EXACT LIMIT OF THE EXPECTED PERIODOGRAM IN THE UNIT-ROOT CASE — João Valle e Azevedo
14/07	INTERNATIONAL TRADE PATTERNS OVER THE LAST FOUR DECADES: HOW DOES PORTUGAL COMPARE WITH OTHER COHESION COUNTRIES? — João Amador, Sónia Cabral, José Ramos Maria
15/07	INFLATION (MIS)PERCEPTIONS IN THE EURO AREA — Francisco Dias, Cláudia Duarte, António Rua
16/07	LABOR ADJUSTMENT COSTS IN A PANEL OF ESTABLISHMENTS: A STRUCTURAL APPROACH — João Miguel Ejarque, Pedro Portugal
17/07	A MULTIVARIATE BAND-PASS FILTER — João Valle e Azevedo
18/07	AN OPEN ECONOMY MODEL OF THE EURO AREA AND THE US — Nuno Alves, Sandra Gomes, João Sousa
19/07	IS TIME RIPE FOR PRICE LEVEL PATH STABILITY? — Vitor Gaspar, Frank Smets , David Vestin

20/07	IS THE EURO AREA M3 ABANDONING US? — Nuno Alves, Carlos Robalo Marques, João Sousa
21/07	DO LABOR MARKET POLICIES AFFECT EMPLOYMENT COMPOSITION? LESSONS FROM EUROPEAN COUNTRIES — António Antunes, Mário Centeno
	2008
1/08	THE DETERMINANTS OF PORTUGUESE BANKS' CAPITAL BUFFERS — Miguel Boucinha
2/08	DO RESERVATION WAGES REALLY DECLINE? SOME INTERNATIONAL EVIDENCE ON THE DETERMINANTS OF RESERVATION WAGES — John T. Addison, Mário Centeno, Pedro Portugal
3/08	UNEMPLOYMENT BENEFITS AND RESERVATION WAGES: KEY ELASTICITIES FROM A STRIPPED-DOWN JOB SEARCH APPROACH — John T. Addison, Mário Centeno, Pedro Portugal
4/08	THE EFFECTS OF LOW-COST COUNTRIES ON PORTUGUESE MANUFACTURING IMPORT PRICES — Fátima Cardoso, Paulo Soares Esteves
5/08	WHAT IS BEHIND THE RECENT EVOLUTION OF PORTUGUESE TERMS OF TRADE? — Fátima Cardoso, Paulo Soares Esteves
6/08	EVALUATING JOB SEARCH PROGRAMS FOR OLD AND YOUNG INDIVIDUALS: HETEROGENEOUS IMPACT ON UNEMPLOYMENT DURATION — Luis Centeno, Mário Centeno, Álvaro A. Novo
7/08	FORECASTING USING TARGETED DIFFUSION INDEXES — Francisco Dias, Maximiano Pinheiro, António Rua
8/08	STATISTICAL ARBITRAGE WITH DEFAULT AND COLLATERAL — José Fajardo, Ana Lacerda
9/08	DETERMINING THE NUMBER OF FACTORS IN APPROXIMATE FACTOR MODELS WITH GLOBAL AND GROUP-SPECIFIC FACTORS — Francisco Dias, Maximiano Pinheiro, António Rua
10/08	VERTICAL SPECIALIZATION ACROSS THE WORLD: A RELATIVE MEASURE — João Amador, Sónia Cabral
11/08	INTERNATIONAL FRAGMENTATION OF PRODUCTION IN THE PORTUGUESE ECONOMY: WHAT DO DIFFERENT MEASURES TELL US? — João Amador, Sónia Cabral
12/08	IMPACT OF THE RECENT REFORM OF THE PORTUGUESE PUBLIC EMPLOYEES' PENSION SYSTEM — Maria Manuel Campos, Manuel Coutinho Pereira
13/08	EMPIRICAL EVIDENCE ON THE BEHAVIOR AND STABILIZING ROLE OF FISCAL AND MONETARY POLICIES IN THE US — Manuel Coutinho Pereira
14/08	IMPACT ON WELFARE OF COUNTRY HETEROGENEITY IN A CURRENCY UNION — Carla Soares

- 15/08 WAGE AND PRICE DYNAMICS IN PORTUGAL
  - Carlos Robalo Marques
- 16/08 IMPROVING COMPETITION IN THE NON-TRADABLE GOODS AND LABOUR MARKETS: THE PORTUGUESE CASE
  - Vanda Almeida, Gabriela Castro, Ricardo Mourinho Félix